BANCA D'ITALIA

RESEARCH DEPARTMENT PUBLIC FINANCE WORKSHOP

Public Debt

Papers presented at the Banca d'Italia workshop held in Perugia, 1-3 April, 2004 Earlier versions of the papers in this volume were presented at the Banca d'Italia workshop on Public Debt held in Perugia, S.A.Di.Ba., on 1-3 April 2004. We wish to thank the staff of S.A.Di.Ba. for their assistance in all aspects of the organization of the workshop.

Previous Public Finance workshops have addressed the following topics:

"Indicators of Structural Budget Balances" (1999)

"Fiscal Sustainability" (2000)

"Fiscal Rules" (2001)

"The Impact of Fiscal Policy" (2002)

"Tax Policy" (2003)

The proceedings may be requested from:

Banca d'Italia Biblioteca Via Nazionale, 91 00184 Roma Italy

The contents of this book do not commit Banca d'Italia. Responsibility for the ideas and opinions expressed rests with the authors of the individual papers.

© 2004 Banca d'Italia

CONTENTS

Daniele Franco (Bank of Italy)					
Intro	duction Fabrizio Balassone, Daniele Franco and Stefania Zotteri (Bank of Italy)	13			
Public	c Debt: A Survey of Policy Issues Fabrizio Balassone, Daniele Franco and Stefania Zotteri (Bank of Italy)	27			
	Session 1 ASSESSING PUBLIC LIABILITIES				
1.	Fiscal Adjustment in EU Countries: A Balance Sheet Approach Gian Maria Milesi-Ferretti* and Kenji Moriyama (*International Monetary Fund) (** University of Michigan)	71			
2.	Public Debt, Contingent Liabilities and Debt "Tolerance": The Case of Colombia Sergio Clavijo Vergara (Bank of Colombia)	97			
3.	A Different View on Public Debt: The Government as Financial Investor Carl Gjersem (Ministry of Finance – Norway)	115			
4.	Public Debt in Emerging Markets: Is It Too High? James Alexander Daniel, Tim Callen, Macro Terrones, Xavier Debrun and Céline Allard (International Monetary Fund)	129			

5. Public Debt Indicators in Latin American Countries: Snowball Effect, Currency Mismatch and the Original Sin Ricardo Martner and Varinia Tromben			
	(United Nations – ECLAC)p.	155	
6.	Fiscal Sustainability and Vulnerability in a Small Open Economy: The Uruguyan Experience Isabel Rial and Leonardo Vicente (Bank of Uruguay)	181	
7.	Fiscal and Generational Imbalances: New Budget Measures for New Budget Priorities Jagadeesh Gokhale* and Karl Smetters** (* Cato Institute) (** University of Pennsylvania and National Bureau of Economic Research)	235	
Com	ments Carl Andreas Claussen (Bank of Norway)	267	
Com	ments Jorge Correia da Cunha (Bank of Portugal)p.	273	
Com	Wents Yoshiko Sato (Bank of Japan)	277	
	Gur to 2		
	Session 2		
	PUBLIC DEBT AND FISCAL RULES		
8.	The Measurement of Government Debt in the Economic and Monetary Union Reimund Mink and Marta Rodríguez-Vives (European Central Bank)	285	

9.	Accounting Regimes and Fiscal Rules Paul Boothe					
	(University of Alberta)p.	315				
10.	The Role of Public Debt in the UK Fiscal Rules Robert Woods					
	(UK Treasury)p.	333				
11.	The Impact of the EU Fiscal Framework on Economic Activity: A Quantitative Assessment					
	Alessandro Turrini* and Jan in 't Veld**					
	(* European Commission and CEPR) (** European Commission)					
	(European Commission)p.	367				
12.	Market-Induced Fiscal Discipline: Is There a Fall-back Solution for Rule Failure?					
	Fabrizio Balassone, Daniele Franco and Raffaela Giordano					
	(Bank of Italy)p.	389				
13.	The Effects of Budget Deficits on Interest Rates: A Review of Empirical Results Thomas Laubach					
	(Federal Reserve Board and OECD)p.	427				
14.	Are Fiscal and Monetary Policies Reflected in Real Yields? Evidence from a Period of Disinflation and Declining Deficit Targets					
	Hedva Ber, Adi Brender and Sigal Ribon					
	(Bank of Israel)p.	447				
15.	Fiscal Policy and Interest Rates in Europe Riccardo Faini					
	(University of Rome)p.	481				
Com	ments					
	Sandro Momigliano (Bank of Italy)p.	513				
Com	ments					
	Jorge Onrubia (Institute for Fiscal Studies and Ministry of Finance – Spain) p.	517				

	Jean-Luc Schneider (Ministry of the Economy – France)p.
	Session 3
	PUBLIC DEBT, AGEING AND FISCAL SUSTAINABILITY
16.	Long-Term Budgetary Implications of Tax-Favoured Private Pension Schemes
	Pablo Antolín, Alain De Serres and Christine De La Maisonneuve (Organisation for Economic Cooperation and Development)p.
17.	Public-Private Partnerships: Implications for Public Finances Marco Cangiano, Richard Hemming and Teresa Ter-Minassian (International Monetary Fund)
18.	Fiscal Institutions and Sustainability of Public Debt in Germany Florian Höppner and Christian Kastrop (Federal Ministry of Finance – Germany)
19.	Public Debt, Sustainability and the Fiscal Policy Framework of the EMU Ilkka Kajaste (Ministry of Finance – Finland)
20.	The Public Debt and the Problem of Population Ageing in Lithuania Violeta Klyvienė (Bank of Lithuania)
21.	Future Public Debt Accumulation and Saving in the United States Charles Steindel

22. The Impact of Different Fiscal Policy Regimes on Public Debt Dynamics				
	Walpurga Köhler-Töglhofer* and Martin Zagler**			
	(* National Bank of Austria)			
	(Vienna University of Economics and Business			
	Administration – Vienna)p.	651		
23.	Sustainability of Public Debt in India: An Assessment in the Context of Fiscal Rules			
	Ranjit Kumar Pattnaik, Anupam Prakash and Biswa Swarup Misra			
	(Reserve Bank of India)	679		
Com	ments			
	Jagadeesh Gokhale			
	(CATO Institute, Washington)p.	737		
Com	ments			
	John Janssen			
	(New Zealand Treasury)p.	745		
Com	ments			
	Karine Hervé (Bank of France)	749		
	Session 4			
	MANAGING PUBLIC DEBT			
24.	Debt Management: A Survey of Theoretical Developments and			
	Innovations in European Practices			
	Guido Wolswijk and Jacob de Haan			
	(* European Central Bank)			
	(** University of Groningen)p.	761		
25.	Debt Management in a Low-Debt Environment: Australia's			
	Experience			
	Blair Comley and David Turvey	703		
	(Australian Treasury)p.	783		

26.	Debt Management in the Czech Republic: Formation in the Nineties and the Current State Ivan Matalík and Michal Slavík				
	(Czech National Bank)p.	829			
27.	Broadening the Approach for Italian Debt Management Maria Cannata, Davide Iacovoni, Stefano Scalera and Manuel Turco (Ministry of the Economy – Italy)	853			
28.	Debt Management in Japan: How to Cope with Interest Rate Risk Mariko Fujii				
	(University of Tokyo)p.	883			
29.	The Monetisation of Japan's Government Debt David E. Lebow (Bank for International Settlements)	901			
30.	Possible Points of Contradiction Between Fiscal Policy and Debt Management Objectives (In Developing Countries) Mariella Nenova and Plamen Kaloyanchev (Bank of Bulgaria)	921			
Com	Aments António Afonso (European Central Bank and ISG/UTL – Technical University of Lisbon)	0.45			
_	• • •	943			
Com	ments Yngve Lindh				
	(Ministry of Finance – Sweden)p.	957			
Com	Ranjana Madhusudhan	0.5			
	(New Jersey Department of Treasury)p.	961			

List of contributors

António Afonso (European Central Bank and ISEG/UTL – Technical University of Lisbon)

Céline Allard (International Monetary Fund)

Pablo Antolin (Organisation for Economic Cooperation and Development)

Fabrizio Balassone (Bank of Italy)

Hedva Ber (Bank of Israel)

Paul Boothe (University of Alberta)

Adi Brender (Bank of Israel)

Tim Callen (International Monetary Fund)

Marco Cangiano (International Monetary Fund)

Maria Cannata (Ministry of the Economy – Italy)

Carl Andreas Claussen (Bank of Norway)

Sergio Clavijo Vergara (Bank of Colombia)

Blair Comley (Australian Treasury)

Jorge Correia da Cunha (Bank of Portugal)

James Alexander Daniel (International Monetary Fund)

Xavier Debrun (International Monetary Fund)

Jacob De Haan (University of Groningen)

Christine De La Maisonneuve (Organisation for Economic Cooperation and Development)

Alain De Serres (Organisation for Economic Cooperation and Development)

Riccardo Faini (University of Rome)

Daniele Franco (Bank of Italy)

Mariko Fujii (University of Tokyo)

Raffaela Giordano (Bank of Italy)

Carl Gjersem (Ministry of Finance – Norway)

Jagadeesh Gokhale (CATO Institute, Washington)

Richard Hemming (International Monetary Fund)

Karine Hervé (Bank of France)

Florian Höppner (Federal Ministry of Finance – Germany)

Davide Iacovoni (Ministry of the Economy – Italy)

Jan in 't Veld (European Commission)

John Janssen (New Zealand Treasury)

Ilkka Kajaste (Ministry of Finance – Finland)

Plamen Kaloyanchev (Bank of Bulgaria)

Christian Kastrop (Federal Ministry of Finance – Germany)

Violeta Klyvienė (Bank of Lithuania)

Walpurga Köhler-Töglhofer (National Bank of Austria)

Thomas Laubach (Federal Reserve Board and OECD)

David E. Lebow (Bank for International Settlements)

Yngve Lindh (Ministry of Finance – Sweden)

Ranjana Madhusudhan (New Jersey Department of Treasury)

Ricardo Martner (United Nations – ECLAC)

Ivan Matalík (Czech National Bank)

Gian Maria Milesi-Ferretti (International Monetary Fund)

Reimund Mink (European Central Bank)

Biswa Swarup Misra (Reserve Bank of India)

Sandro Momigliano (Bank of Italy)

Kenji Moriyama (University of Michigan)

Mariela Nenova (Bank of Bulgaria)

Jorge Onrubia (Institute for Fiscal Studies and Ministry of Finance - Spain)

Ranjit Kumar Pattnaik (Reserve Bank of India)

Anupam Prakash (Reserve Bank of India)

Isabel Rial (Bank of Uruguay)

Sigal Ribon (Bank of Israel)

Marta Rodríguez-Vives (European Central Bank)

Yoshiko Sato (Bank of Japan)

Stefano Scalera (Ministry of the Economy – Italy)

Jean-Luc Schneider (Ministry of the Economy – France)

Karl Smetters (University of Pennsylvania and National Bureau of Economic Research)

Charles Steindel (Federal Reserve Bank of New York)

Michal Slavík (Czech National Bank)

Teresa Ter-Minassian (International Monetary Fund)

Macro Terrones (International Monetary Fund)

Varinia Tromben (United Nations – ECLAC)

Manuel Turco (Ministry of the Economy – Italy)

Alessandro Turrini (European Commission and CEPR)

David Turvey (Australian Treasury)

Leonardo Vicente (Bank of Uruguay)

Guido Wolswijk (European Central Bank)

Robert Woods (UK Treasury)

Martin Zagler (University of Economics and Business Administration, Vienna)

Stefania Zotteri (Bank of Italy)

FOREWORD

Daniele Franco

This volume collects the papers presented at the Sixth Banca d'Italia Public Finance Workshop, held in Perugia on 1-3 April 2004.

The workshop aimed at providing an overview of the theoretical and empirical work concerning public debt. It included analytical papers examining the definition, the measurement, the role and the effects of public debt. It also included papers tackling policy issues from different angles and different national perspectives.

The workshop allowed experts from central banks, ministries and economic institutions of several countries, the main international economic organisations and the academic world to discuss the economic effects of public debt and policy options.

The papers illustrate the complexity of the definition of public liabilities. They examine the effects of public debt on saving, financial markets and intergenerational redistribution. They consider the role of public debt in the design and implementation of fiscal policy rules at the national and European levels. They evaluate the role of public debt in a context of demographic ageing. They examine the evolution of debt management.

The various methodological approaches of the papers highlight the wide variety of research in the field of public debt analysis. Some papers draw from theoretical models, others from empirical investigations and others from the policy debate.

Banca d'Italia thanks all institutions which contributed to the success of the initiative and all experts who provided research papers and who came to Perugia to take part in the discussions.

This volume develops further the analysis of fiscal policy issues carried out in the proceedings of the previous workshop, which were devoted to *Indicators of Structural Budget Balances* (November 1998), *Fiscal Sustainability* (January 2000), *Fiscal Rules* (February 2001), *The Impact of Fiscal Policy* (April 2002) and *Tax Policy* (April 2003).

^{*} Bank of Italy, Research Department.

INTRODUCTION

Fabrizio Balassone, Daniele Franco and Stefania Zotteri*

Public debt has long been at the centre of the fiscal policy debate.

Public borrowing is a powerful tool of economic policy. Via borrowing governments can affect the allocation of resources, economic activity and the distribution of income and wealth. Governments can use borrowing to meet exceptional events, to counter economic downturns and to expand infrastructures.

On the other hand, the misuse of public borrowing can have significant and long-lasting implications. The stock of public debt influences economic decisions and expectations. It constraints the room for manoeuvre of fiscal policy. It can require painful adjustments or it can cause inflation or default.

This dichotomy has prompted a long debate on the role and the limits of public debt. Economists, philosophers and policy makers have highlighted many, sometimes radically different, views. The debate has reflected the diverging opinions concerning the role of the state, the objectives of policy makers and the effectiveness of fiscal policy. While taking different nuances over time, it has also shown a number of constant features.

The debate on public debt has been shaped by different experiences concerning its use in policy making. History has seen numerous episodes of debt accumulation driven by different economic and political factors. Debt decumulation via consolidation, inflation or default has frequently proved economically problematic and has produced significant political consequences.

The papers collected in this volume consider the main strands of the current debate on public debt. Some papers explore the issue of the definition of public liabilities. They highlight the complexity of the issue, especially in view of the assessment of fiscal sustainability and the introduction of formal fiscal rules. Some papers evaluate the role of public debt in emerging market economies. They point to the specific problems raised by debt accumulation in these countries. Some papers examine the role of public debt in the design of fiscal rules, particularly in the EU context. They also consider the relation between deficits and interest rates and evaluate the possibility to ensure fiscal discipline via market mechanisms. Some papers examine the interaction between public debt and demographic ageing. They point to the need to consider the long-term implications of current policies and to the constraint posed by high debt ratios. Finally, some papers examine debt management. They highlight the trade-offs between different objectives and the progress in the techniques.

E-mail: FBalassone@imf.org, daniele.franco@bancaditalia.it, stefania.zotteri@bancaditalia.it

^{*} Bank of Italy, Research Department.

The papers presented at the workshop were allocated in four sessions which are mirrored by the sections in this volume. Section 1 examines the definition and measurement of public debt. Section 2 considers the role of public debt in the design of fiscal rules. Section 3 examines debt in the context of demographic ageing. Section 4 deals with public debt management.

The introductory paper provides a concise overview of the main issues in the debate over public debt. After reviewing the reasons for debt from different analytical perspectives, the paper turns to the definition and assessment of debt sustainability and discusses the implications of high debt levels for the macroeconomic performance of the economy. The final part of the introductory paper considers market-based and rule-based mechanisms to control debt growth and examines the issues arising in the measurement of public liabilities as well as the implications of fiscal rules for debt management.

1. Assessing public liabilities

The papers in Section 1 address the twin issues of which indicators and which methods should be used to assess the sustainability of public debt. They do so from very different perspectives, ranging from the effectiveness of fiscal rules and of the policy making process in general, to the specifics of emerging markets and to the role of government assets.

Milesi-Ferretti and Moriyama use a balance sheet approach to evaluate to what degree changes in the size of gross public debt in EU countries reflect corresponding changes in holdings of government assets or underlying increases in net worth. They describe a number of fiscal measures that improved the fiscal accounts subject to the Maastricht criteria but with no durable impact on public finances as a whole. They find that the correlation between changes in government liabilities and changes in government assets was positive and strong over 1992-97 and that it weakened over 1997-2002. They point out that this is consistent with the notion that up to1997 governments contained the rise in the public debt ratio (or reduced it) also by decumulating government assets, regardless of the impact on net worth, in order to comply with Maastricht fiscal criteria. This effect waned over 1997-2002 once the consequences of resilient debt-to-GDP ratios became less punishing.

Clavijo Vergara analyses the dynamics of Colombia's public and external debt over the period 1997-2003. He argues that a proper assessment of debt sustainability requires that the cost of serving the debt is computed on a gross basis (*i.e.* including the interest payments on intra-governmental debt) and that contingent liabilities, like pension obligations and public guarantees, are taken into account. He remarks that, in spite of the pressure from both international organisations and market players, these two conditions are not usually met. He finds this particularly worrisome in the light of recent evidence which shows that recognition of contingent

liabilities in emerging markets, along with interest rates and exchange rate developments, accounts for the bulk of public debt deterioration.

The paper by Gjersem notes that in Norway revenues from the petroleum sector have reached their peak and will slowly dwindle over the coming decades while, at the same time, population ageing will put an upward pressure on pension and health care costs (as in most other OECD countries). Against this background, the author examines the case for pre-funding and saving in the Norwegian public sector and discusses the working of the Petroleum Fund. The latter was established in 1990 to enhance the transparency of the spending of petroleum revenues and to finance the accumulation of assets to be used to meet long-term challenges related to demographic development. Gjersem describes the structure of the Fund, its investments and financial results.

Daniel, Callen, Terrones, Debrun and Allard apply a number of different approaches to assess fiscal sustainability in emerging markets. Their analysis suggests that for these countries the sustainable level of public debt is often quite low. They discuss the heavy policy agenda that confronts policies to reduce public debt and to cushion against the risks that high debt presents. These policies include: a) reforms to strengthen and broaden the tax base; b) better control of expenditures during economic upswings; c) improving the credibility of fiscal policy; d) reducing exposure to exchange rate and interest rate movements; e) structural reforms to boost growth prospects; f) addressing the risks from contingent and implicit liabilities.

Martner and Tromben argue that in Latin American countries the threats to debt sustainability originate in the pro-cyclical bias of fiscal policies. The threats are reinforced by the difficulty that these countries experience in borrowing with long-term maturity domestically or in borrowing in national currency on international markets. This double mismatch (in terms of maturity and of currency) intensifies the uncertainty of public debt service, thus lowering credit ratings and further increasing the cost of debt (also via exchange rate depreciation). In these circumstances, if the reaction of fiscal policy is not timely and quantitatively adequate (*i.e.* if debt-stabilising fiscal primary surpluses cannot be generated in the short term), the debt can take an unsustainable path. Martner and Tromben also discuss proposals advanced in the literature that may lessen these problems.

Rial and Vicente argue that in the case of a small emerging economy facing recurrent shocks of significant magnitude, the analysis of fiscal sustainability cannot be based on the dynamics of the debt level only. The structure of debt by currency, interest rate, maturity, type of instrument, etc. is most important. Using indicators that quantify and evaluate the risks related to the debt structure, they show that in Uruguay, in spite of the low levels of debt-to-GDP ratio observed at the beginning of the Nineties, vulnerability to shocks in debt determinants was very high even at that time. They also note that, for the future, an increase in the share of domestic currency denominated debt would be necessary to ensure debt sustainability.

Gokhale and Smetters argue that traditional budget measures are especially ill-suited to monitor fiscal sustainability in the context of growing social insurance programs. Focussing on the USA, they suggest that federal budget agencies should begin reporting different indicators which can give a better sense of the fiscal situation and of the implication of alternative means of correcting it. Specifically, they suggest to refer to the Fiscal Imbalance Indicator and to the Generational Imbalance Indicator. The former is the difference between all projected federal receipts and the sum of the current federal debt held by the public plus the present value in today's dollars of all projected federal non-interest spending. The latter measure indicates how much of this imbalance is caused by past and present generations.

The comments by Claussen focus on the four papers on public debt in emerging markets. He points out that there is no simple rule for determining whether a government's debt is sustainable or not and argues that while the approaches considered by the papers may give a useful account of the situation of emerging markets as a group, they should be used with caution when assessing the situation of each country separately. He suggests that a good measure of sustainability has to be based on the notion that the borrowing and default decision of a government is the result of a dynamic political economy game. While acknowledging the difficulties involved, he argues that such a measure could be built starting from a detailed model of the costs of default.

Correia da Cunha reconsiders the analysis in the paper by Milesi-Ferretti and Moriyama. He acknowledges that net worth is superior to gross debt as a fiscal indicator as it is not affected by re-compositions in general government assets and liabilities. However, he underlines the high informational requirements for a proper computation of net worth (specifically with reference to the value of the public capital stock and to estimates of its depreciation). He also points out that there are measures which can exert a negative effect on sustainability without affecting net worth, such as an increase in future pension payments or in payments related to the construction of infrastructures under public-private partnership schemes. To limit recourse to creative accounting, Correia da Cunha suggests, first, monitoring more than one indicator and, second, reinforcing the independence of the national statistical institutes and the ability of parliaments to follow fiscal developments.

Satou notes that while assessing public liabilities is not an easy task, all the papers in the first session have produced interesting results. She examines separately the papers dealing with Latin America and those concerning European countries. She finds that the main points made in the first set of papers concern: (a) the inadequacy of the primary surplus run by most Latin American countries with respect to what is necessary for debt sustainability; (b) the relevance of hidden liabilities; and (c) the contribution of external factors. Concerning Europe, she points out that while fiscal conditions are good as compared to emerging market economies, many issues remain to be studied. She also warns that the interpretation of net worth figures should take into account developments in asset prices and

argues that financing the deficit via revenues from assets can determine distortions in the allocation of resources.

2. Public debt and fiscal rules

Section 2 includes eight papers which focus on different issues concerning public debt and fiscal rules. More specifically, the papers deal with three relevant topics: the effectiveness of fiscal rules in keeping debt under control; the relationship between fiscal rules and choices concerning debt measurement and accounting criteria; the effects of fiscal rules on macroeconomic variables, namely interest rates and economic activity.

The paper by Mink and Rodríguez-Vives focuses on debt measurement issues with a specific reference to euro-area countries. Two debt measures are discussed: the debt relevant for European fiscal rules (the Excessive Deficit Procedure debt measure) and the ESA95 debt. The former is analysed by breaking it down by financial instrument, holder, government sub-sector, maturity and currency. Differences between the two measures reflect differences in instrument coverage, in the treatment of accrued interests and in the valuation method applied. The paper also briefly refers to issues concerning the possible inclusion of implicit liabilities in the debt statistics and the possible computation of a net measure of debt.

Boothe contributes to the literature that focuses on the impact of budgetary institutions on fiscal outcomes by examining the interaction between accounting regimes and fiscal rules. The paper builds and calibrates a model through which cash and accrual accounting regimes are compared and their interaction with different fiscal rules is analysed. The paper does not take a position on the suitability of cash versus accrual regimes. It rather looks for the circumstances under which governments may choose to move from an accounting system to the other. The author stresses that rules which discourage deficit financing are harder to comply with under cash accounting than under accrual accounting.

Woods analyses the role of public debt within the UK fiscal rules. The analysis is framed in the context of both the theoretical literature on fiscal sustainability and of the UK debt history. The UK government has two rules: the golden rule and the sustainable investment rule. The paper focuses on the latter, which requires public sector net debt as a proportion of GDP to be held at a stable and prudent level over the economic cycle. More specifically, net debt should be maintained below 40 per cent of GDP over the economic cycle. First, the paper discusses the theoretical basis underlying the role of the debt within the UK fiscal framework. Then it assesses the degree of sustainability of UK public finances. The paper concludes that, on the basis of current policies and under a range of reasonable assumptions, the UK public finances seem sustainable in the long term.

The paper by Turrini and in't Veld focuses on the impact of the European fiscal rules on economic activity. Their investigation concerns euro-area countries and it is in two steps. First, they provide an estimate of the fiscal outcomes that

would have prevailed in the absence of the European fiscal framework. Without fiscal rules, euro-area countries would have ended up with higher primary deficits and higher debt. Secondly, these counterfactual public finances outturns are plugged into the European Commission's econometric model to assess their effects on macroeconomic variables. The authors find that, especially when the effects of worsening in public finance variables on risk premia are taken into account, the positive GDP effects are short-lived.

In the light of the ongoing discussion concerning the effectiveness of the European fiscal rules, Balassone, Franco and Giordano reconsider the debate which took place before the approval of the Maastricht Treaty by investigating whether market mechanisms can be relied upon as a fall-back solution in case of rule-failure. First, the conditions for an effective market solution are examined and the European institutional framework is assessed against them. Second, the relationship between fiscal performance, credit rating and interest rates is discussed with reference to both theory and practice in EMU countries. Finally, governments' sensitivity to market signals is analysed. The findings point to a significant, though small, reaction of interest rates to fiscal imbalances and to a slow government response. The authors conclude that market mechanisms cannot be relied upon for replacing fiscal rules even if they can play an important complementary role.

Laubach provides a critical review of the empirical literature that focuses on the effects of budget deficits on interest rates. The main empirical problem in estimating this relationship is to control for other factors determining real interest rates. In particular, the simultaneous response of monetary policy and automatic stabilisers can in principle mask the effect of discretionary fiscal policy on interest rates. The paper stresses that simple regressions of current interest rates on current budget deficits yield ambiguous results which are consistent with the view that endogeneity problems in such regressions are pervasive. The paper considers different solutions for coping with these problems. It shows that when endogeneity problems are properly accounted for, mainly by adequately considering expectations of both deficits and interest rates, available studies tend to find strong evidence that increases in budget deficits raise interest rates.

Also the paper by Faini focuses on the relationship between fiscal policy and interest rates. With reference to the European context, the paper addresses two issues. First, it investigates whether national fiscal policies either affect country specific spreads or the average level of euro-area interests rates. Only in the latter case are there significant externalities on the area from national misbehaviour. Second, it discusses whether this negative externality is likely to be more important for high-debt countries. The empirical analysis shows that while national fiscal policy has a limited impact on spreads, it has a more important impact on the area interest rates. Moreover, lack of fiscal discipline in high debt countries has a stronger impact on interest rates. Nevertheless, the problem is not high debt *per se*, but high debt together with lack of fiscal discipline.

Ber, Brender and Ribon study the effects of fiscal and monetary policy on bond yields with reference to the Israeli experience during the Nineties. They find

that both monetary and fiscal policy affect the money market yields. On the one hand, fiscal policy has a direct impact on the money market via the expected deficit and an indirect one via inflation expectations. The former mechanism is relevant only when the effects of the cycle are accounted for and it is larger on long-term than on short-term yields. On the other hand, monetary policy has a marked direct impact on short-term yields, but also on long-term ones. The paper finds that the monetary policy effects on the money market dominate and notes that this result may be due to the disinflationary policy implemented in Israel during the Nineties.

In commenting the papers of this section, Onrubia points to the three issues they cope with: the effectiveness of fiscal rules in controlling public debt; the measurement of public debt; the relationship between public debt, budget deficits, interest rates and inflation. With reference to the effectiveness of fiscal rules, he stresses the importance of transparency and the role that institutions can play in achieving it. He also underlines the importance of considering the role played by uncertainty within long-term sustainability analysis, an issue which is considered in Wood's paper. Finally, Onrubia mentions the importance of institutions as a support not only to formal fiscal rules but also to appropriate accounting practices.

Schneider reconsiders the analyses and the evidence provided by the papers of this section in order to answer two questions. Does the market respond to fiscal laxity? Is the market response adequate? He believes that the market responds to fiscal deficits and that there are economic reasons why empirical analyses tends to underestimate this response. Indeed, by using macro-econometric models as a benchmark, he stresses that market response seems adequate. If so, why should fiscal rules be useful? Schneider gives many reasons, among which are the timing of the market response, the importance of anchoring expectations and the possibility that markets may misperceive fiscal outcomes.

Also Momigliano focuses attention on the relationship between fiscal policy and interest rates. He notes that there are two main problems to face when trying to assess empirically the impact of fiscal policy on interest rates. First, cyclical conditions tend to affect simultaneously both the budget balance and interest rates. This endogeneity tends to distort the estimates of the coefficient of the fiscal variables. Secondly, the budget balance may affect nominal interest rates either via expectations or via effects on the real interest rates. Distinguishing between these two components may be relevant.

3. Public debt, ageing and fiscal sustainability

Section 3 includes eight papers concerning medium and long-term issues. Two of them are about the assets and liabilities implicit in private pensions and public-private partnership. Four papers concern the role of public debt in the design of fiscal policy frameworks aimed at ensuring fiscal sustainability. The ageing context is also considered. The other two papers focus on the effect of debt on

saving decisions and on the relationship between public debt and fiscal policy regimes.

Antolín, de Serres and de la Maisonneuve examine the long-term budgetary implications of the tax incentives provided to private pension schemes. The paper considers seventeen OECD countries and focuses on schemes that generate tax deferral. It estimates the flows of budgetary costs and revenues related to the schemes over the period 2000-50 and the present value of the implicit fiscal asset of governments. According to the paper, tax-favoured schemes are likely to remain costly over the period considered in spite of the increase in revenues resulting from population ageing. Budgetary costs would be smaller if tax incentives were to induce additional savings. The authors note that these results do not question the support granted in many countries to private pension schemes. *Inter alia*, this support allows government to shift sizeable tax revenues to the period in which the impact of ageing on public budgets will peak.

The paper by Cangiano, Hemming and Ter-Minassian evaluates the implications for public finances of public-private partnerships (PPPs). An increasing number of countries have introduced schemes where the private sector supplies infrastructure assets and services. While these schemes can increase the supply of infrastructure and reduce costs, government guarantees can be a major source of fiscal risk. The paper examines the main features of PPP schemes, the underlying economic problems and the conditions that can make the recourse to PPPs efficient. The authors also examine the institutional features of PPPs: the legal framework, risk transfer, fiscal accounting and reporting. The paper concludes by pointing to the need to carefully assess the budgetary risks associated with PPPs, to ensure appropriate accounting standards and to strengthen disclosure requirements concerning the underlying risks and contingent liabilities.

Höppner and Kastrop examine the role of fiscal institutions and budgetary procedures in Germany with respect to the objective of safeguarding sound public finances in the long run. The paper analyses the main factors underlying debt developments in Germany in recent decades. It also examines budgetary prospects, pointing to the increasing share of public expenditure which depends on entitlements which cannot be modified over the short-term and to the spending increases stemming from the ageing process. The paper then moves to considering the role of medium-term planning in the German budget and the current approach to long-term issues. In this context, Höppner and Kastrop evaluate some possible reforms. In particular, they consider the use of indicators concerning the quality of public finances, such as the "Public expenditure for growth and sustainable development". In view of the federal structure of the country, they also examine the possibility of introducing a National Stability Pact.

The paper by Kajaste examines the role of long-term sustainability and debt issues in the European Union fiscal framework. It laments the lack of an explicit and operational link between the Stability and Growth Pact and the sustainability of public finances and it notes that this aspect may have contributed to the recent unsatisfactory outcomes in the implementation of EU fiscal rules. The paper

examines the steps taken by the European Council and the European Commission to increase the focus on sustainability and debt issues in EU fiscal surveillance. It also examines the role that these issues have played in the surveillance concerning some potentially problematic countries. Finally, the paper examines some further steps that could be taken at the national and European level to enhance the role of long-term fiscal considerations.

The paper by Klyvienė examines the prospects for budgetary policy in Lithuania. It evaluates the recent evolution of public debt and provides estimates concerning contingent liabilities. The paper also examines demographic trends and points to a sizeable increase in the dependency ratio. It also presents alternative scenarios concerning the balance of the State Social Insurance Fund and their implications for public debt. The paper shows that public debt is currently rather low and the conditions of the Fund in the coming years are expected to be rather favourable. However, demographic trends are likely to determine sizeable deficits and an increase in the debt level. The deficit of the State Social Insurance Fund will also be increased by the introduction of the second funded pillar. Klyvienė points to the need to introduce further pension reforms, such as extending the coverage of the system and further increasing retirement age.

The paper by Steindel examines the impact on saving decisions of the expected future debt growth in the USA arising from the increase in pension and health spending. After having considered the long-term outlook of the main social programmes, pointing to the expected strains, the paper evaluates the connections between consumer behaviour and entitlement programs. Steindel argues that the provision of public pensions does not seem to depress savings and capital formation. This may depend on the frequency of changes in the rules concerning public entitlements: people do not expect that social expenditure growth will continue unchecked and discount the effects of the likely future cuts in entitlements. This suggests that reform measures need not be large in the short-term. Finally, the paper evaluates some alternative reform proposals for the US public pension and health care schemes.

Köhler-Töglhofer and Zagler investigate the impact of compositional effects on public debt dynamics under different fiscal policy regimes. The analysis moves from the literature on the role of compositional effects for the success of fiscal consolidation episodes to a broader set of policy regimes. It also considers the role of the macroeconomic environment. On the basis of evidence drawn from a large sample of countries over the period 1960-2002, the paper finds that compositional effects do not show statistically significant differences across policy regimes. It also shows that individual revenue and primary expenditure categories have an important impact on debt dynamics. A reduction in government wage consumption exhibits the strongest dampening impact on debt dynamics. In contrast to the literature, the paper finds that an increase in government revenues leads to a persistent decline in debt ratios.

The paper by Pattnaik, Prakash and Swarup Misra evaluates the sustainability of fiscal policy in India. It reviews the literature on fiscal sustainability highlighting

the specific Indian contributions. It evaluates the Indian experience with fiscal rules in the context of the international debate. The paper also examines the fiscal situation in India, pointing to the efforts to ensure sustainability but also to the relatively large budget deficit. Fiscal sustainability is assessed on the basis of different approaches. The paper notes that the cost of debt in India is lower than nominal GDP growth; nevertheless the debt-to-GDP ratio is likely to increase because of the high deficit. The authors point to the need for further fiscal consolidation. They suggest to restrain spending and they stress the need for increasing public revenues.

In commenting the papers, Gokhale argues that the current generational stance of US fiscal policy provides dependable resource transfers towards retirees and prompts higher consumption. Large and credible transfers induce more consumption by the cohorts that receive them and, all else being equal, reduce national saving. Hence, reforms that terminate such transfers would stem the decline in national saving and a larger share of US domestic investment would be financed out of domestic saving. Output growth would be faster. Gokhale also notes that standard deficit and debt levels are neither necessary nor sufficient as measures of the real impact of fiscal policies on the economy. Implicit debt levels can be changed independently and can influence real economic outcomes.

In commenting the papers, Janssen focuses on two aspects: PPPs and long-term sustainability. Concerning the former issue, he notes that it is important that governments reduce costs and transfer risk and he highlights the conditions for this. Concerning long-term sustainability, he points to the experience of countries, such as Australia, New Zealand and the United Kingdom, which have incorporated specific long-term fiscal reporting requirements into the frameworks guiding their fiscal policy. He stresses the importance of fiscal reporting and of transparency.

Hervé outlines the impact of demographic trends on public expenditure for pensions, health and long-term care. She also comments on the impact of ageing on labour market and saving ratios. Finally, she evaluates some policy actions that may tackle the problems posed by ageing. In particular, she argues in favour of labour market policies aimed at increasing the supply of labour and of measures improving the budgetary situation.

4. Managing public debt

Section 4 includes seven papers focussing on debt management. They refer to the experience of specific countries such as Australia, Bulgaria, the Czech Republic and Japan and those belonging to the European Economic and Monetary Union. Different issues are covered: convergence in debt management strategies, the relationship between fiscal and monetary policy, debt management policy objectives and trade-offs.

Wolswijk and de Haan survey recent developments in debt management in Europe against the background of the introduction of the euro and of declining

government debt ratios. They find a strong convergence in debt management strategies in the euro area in terms of debt maturity, issuance of foreign currency denominated debt and use of more complex instruments than before EMU (interest rate swaps and inflation-indexed bonds are prominent examples). Convergence also seems to apply to the organisation of debt management, with a tendency to make debt management offices more independent. The authors note that some divergences remain, reflecting differences in the size of deficits and debts, as well as an increasing willingness to innovate and to attract investors' attention.

The paper by Comley and Turvey analyses debt management in Australia. The country's fiscal position became increasingly strong from the second half of the Nineties. General government net debt fell from almost 20 per cent of GDP to less than 4 per cent between 1996 and 2003. Concerns that this reduction would negatively affect financial market efficiency (mainly due to the higher cost of managing interest rate risk in the absence of a Treasury bond futures market) led the Australian government continuing to issue debt. The authors stress that this has implied that issuance policy is less and less targeted to achieving desirable cost and risk characteristics for the Government.

Matalík and Slavík examine the evolution of debt management in the Czech Republic. As with many other features of the Czech economy, changes have been dramatic in this field in connection with the transition from central planning to market mechanisms. The paper outlines the shifting policy focus: from the need to establish treasury bills and government bond markets (and to ensure and promote their liquidity) in the early stages of the process to the fine tuning of debt risk management in later years. The latter issue became relevant in connection with the increase in the outstanding debt which induced policies aimed at decreasing government dependence on the domestic bond market.

The paper by Cannata, Iacovoni, Scalera and Turco examines the trade-off between cost and risk faced by countries trying to minimise the burden of servicing a high debt in the context of the European monetary union. The authors present both a model for optimal portfolio selection and one for forecasting primary balances. They argue that the high number of stochastic factors at play and the long-term horizon that needs to be taken by the debt manager combine to make the task especially difficult. They stress that while formal models can help in fulfilling the task, no model can capture all the relevant elements. Therefore, the debt manager's experience must also be relied upon in the implementation of a strategy.

The paper by Fujii provides a throughout description of the features of the Japanese public debt in order to analyse interest rate risk. More specifically, given the high debt level, the paper aims at assessing the impact on public finances of future increases in interest rates by using a stochastic simulation. In the Japanese case no foreign-currency denominated bonds are issued and the share of floating-rate notes and bonds is negligible, so the most relevant policy option concerns the debt maturity structure. The paper concludes that a short-maturity pattern strategy increases the size of market risk even in the short run. The different risk implied by

a short-maturity structure as opposed to a long-maturity structure gets bigger as the simulation period extends. Quantitative evaluations are provided.

Also the analysis by Lebow concerns Japan. Given the high public debt levels, it focuses on the fiscal implications of the monetary expansion undertaken since the early nineties and those of a possible reversal of such a policy in case inflation picks up. The paper argues that for analysing fiscal policy, the consolidated government and central bank net debt is the appropriate aggregate to be considered. According to this view, even if the ratio of public debt to GDP is high, there are important offsetting elements. First, the central bank has already monetised to a relevant extend this debt. Moreover, in case the monetary expansion is allowed to generate temporary inflation, the consolidated debt ratio would fall further. Finally, even in the case of a reversal of the monetary stance, there would be consolidated debt levels lower then generally recognised if small price increases are allowed.

Nenova and Kaloyanchev discuss possible conflicts between the targets set by budgetary policy and those set by debt management in developing countries. These countries are typically characterised by low income levels, poor infrastructure and fragile confidence in policies commitment. In order to cope with these problems budgetary policy is likely to lead to deficit financing and to debt accumulation. In this case the scope of debt management operations narrows and this narrowing positively depends on the speed of debt accumulation as it influences the dynamics of the risk premium on debt. In order to bring debt dynamics under control, the government has to inverse its priorities and to privilege debt management objectives. The authors stress that even if the fiscal adjustment takes place, it may take a long time before the risk premium reaches low levels (especially if there has a been a default on debt). These findings are confirmed with reference to the Bulgarian experience.

Afonso separately sums up and reviews every single paper included in this section. He notes that in principle there can be conflicts between debt management and fiscal policy objectives and it should be so. He supports the reasons for having government debt put forward in the paper concerning Australia. In addition, he stresses the risks associated with the presence of the so-called hidden debt which typically characterises transition economies. Finally, he suggests that the problems stemming from the high Japanese debt ratios should be tackled directly by targeting sizeable budgetary surpluses.

Lindh notes that the contributions of this section can be split into four groups corresponding to the geographic areas they focus on: Australia, euro area, European transition economies and Japan. He relates the paper about Australia, a very low debt country, to those concerning euro-area countries. Indeed, the latter countries should aim at a medium-term objective of close to balance or in surplus which would lead them to have diminishing and low debt levels. With reference to European transition economies, Lindh stresses that the issue of interaction (and possibly coordination) between fiscal, monetary and debt management policies is crucial.

Madhusudhan's comments touch upon the topics covered by the papers of this section by providing evidence concerning the USA. She recalls that the Federal Reserve Bank has no significant direct role in debt management. Moreover, at the federal level debt managers are primarily concerned with maintaining liquidity and efficiency in the financial markets rather than with keeping interest costs under control via swap operations. Finally, contrary to most countries considered in this section, in the USA issuances with relatively short maturities are relevant and a relatively large share of debt is held by foreign investors.

PUBLIC DEBT: A SURVEY OF POLICY ISSUES

Fabrizio Balassone, Daniele Franco and Stefania Zotteri*

Introduction

The role, the limits and the effects of public debt have long been at the core of the fiscal policy debate. Public debt affects the allocation and distribution of resources and the stabilisation function of government. It reflects decisions taken by previous generations and it constrains those of future generations. History has seen numerous episodes of debt accumulation driven by different economic and political factors. Debt decumulation via consolidation, inflation or default has frequently proved economically problematic and has produced significant political consequences.

The debate on public debt has involved economist, philosophers and policy makers, and has highlighted many, sometimes radically different, views.

Ricardo refers to the debt as "... one of the most terrible scourges which was ever invented to afflict a nation", as "... a system which tends to make us less thrifty, to blind us to our real situation". He feared that the citizen initially "deludes himself with the belief, that he is as rich as before" and then, faced with the taxes levied to pay for the debt, is tempted "... to remove himself and his capital to another country, where he will be exempted from such burthens". Smith argued that government borrowing would deprive society of resources which could be invested more productively. He also noted that beyond a certain threshold debt inevitably leads to national bankruptcy.

However, classical economists were also well aware of the necessity of allowing borrowing in certain circumstances and of its usefulness in others. Building on such awareness, gradually the idea gained consensus that the public debt need not be repaid as it can be refunded and that "the problem of the debt burden is a problem of an expanding national income. How can a rapidly rising income be achieved?" (Domar, 1944, p. 166).

This paper aims at providing a concise overview of the main issues surfacing the debate over public debt. In Section 1 we review the main economic factors explaining the existence of debt from three perspectives: public finance, monetary policy and political economy. Section 2 takes a positive point of view and is dedicated to the definition of debt sustainability and to the analytical tools available

Bank of Italy, Research Department. E-mail: FBalassone@imf.org, daniele.franco@bancaditalia.it, stefania.zotteri@bancaditalia.it

See Tanzi and Fanizza (1994) and Masson and Mussa (1995).

² In Sraffa (1951), p. 197.

³ In Sraffa (1951), p. 247-8.

to undertake its assessment. Section 3 discusses the implications of high debt levels for the macroeconomic performance of the economy. Section 4 turns normative and considers market and rule-based mechanisms to control debt growth. Section 5 and 6 are devoted to more technical aspects concerning both analysis and policy: the former examines the issues arising when measuring public liabilities, the latter considers how fiscal rules and indicators can affect government debt management.

1. Why public debt?

In this Section we review the main economic factors explaining the existence of debt from three perspectives: public finance, monetary policy and political economy. The first two are normative, they suggest reasons why debt can be a useful tool; the third is positive and deals with factors driving recourse to deficit finance in practice and with the ensuing risks of excessive debt accumulation.

1.1 Public finance

The precept of a balanced budget, drawn by way of analogy from family finances, found a widespread endorsement well into the XX century as is witnessed by Pigou's 1929 writing: "in normal times the main part of a government's revenue is required to meet regular expenditure that recurs year after year. There can be no question that in a well-ordered State all such expenditure will be provided for out of taxation, and not by borrowing. To meet it by borrowing ... would involve an ever-growing government debt and a corresponding ever-growing obligation of interest. ... The national credit would suffer heavy damage; ... This thesis is universally accepted" (1929, p. 233).

However, even in family finance borrowing is not necessarily evil. Even classical advocates of the balanced budget were aware of the necessity of allowing borrowing in certain circumstances and of its usefulness in others. Therefore economists have had a hard job in trying to specify under what circumstances exceptions to the balanced budget rule were to be allowed and in striking the right balance between the risk of missing opportunities as a consequence of the constraint and the instability and wastes possibly caused by its removal.

Indeed the need for exceptions is clearly recognised by Pigou (1929). He deemed it to be plain that when "non-remunerative government expenditures on a wholly abnormal scale have to be undertaken, as in combating the consequences of an earthquake or to meet an imminent threat of war ... to collect what is required, and required at a very short notice in these conditions, through the machinery of taxation is politically and administratively impracticable" (p. 39; italics ours). He also argued that concerning "government expenditure devoted to producing capital equipment ... the fruits of which will subsequently be sold to purchasers for fees ... it is generally agreed that the required funds ought to be raised by loans. ... Upon this matter ... there is no room for controversy" (p. 36; italics ours). Finally, he

notes that "...since changes in taxation always involve disturbance, to keep the rates of taxation as nearly as possible constant from year to year ... it may be desirable ... to arrange a budget so that good and bad years make up for one another, a deficit in one balancing a surplus in another" (p. 35; italics ours).

However, allowing for exceptions to the balanced budget rule may open room for opportunistic behaviour (see Section 1.3). The theoretical soundness of the principles restricting government borrowing to fixed capital investment (the so called golden rule) can be questioned. The accumulation of debt either during recessions or in periods characterised by extraordinary negative events is justifiable not only to minimise distortions⁴ but also for stabilisation purposes. The feasibility of an effective policy in this respect is not unproblematic.

Concerning the golden rule, clearly there are current expenditures, such as those increasing human capital, that can give a relevant contribution to growth as "indirect revenue need not come through a durable good" (Steve, 1972, p. 164; our translation). The dual budget (*i.e.*, the separation of current and capital operations) may result "... in a preference for expenditures on physical assets rather than greater spending for intangibles such as health or education" (Colm and Wagner, 1963, p. 125). Thus, "the need for a return, either in the limited financial sense or in the broader context of the social return, is a view that needs to be applied over a wider spectrum of public expenditures and not confined to capital budget only" (Premchand, 1983, p. 296). However, the inclusion in the capital account (which can be financed through debt) of all expenditures contributing to human capital would imply high deficit levels.

Deficit financing for stabilisation policy found early applications in Sweden and in the USA. In 1937 Sweden reformed its budget rules and abandoned the annual balancing. In Lindbeck's (1968) account, the Swedish reform was based on the idea that "in normal times the capital budget should be financed by loans whereas the current budget should be financed by taxes. In boom periods the current budget should, however, be overbalanced, hence part of the capital budget would be financed by taxes; in recession the current budget should be underbalanced, hence partly financed by loans" (p. 33). Hansen (1941) explains how in the USA, "President Roosevelt [...] expressed the hope that in times of prosperity current revenues would so far exceed ordinary expenditures as to produce 'a surplus that can be applied against the public debt'... The extraordinary expenditures, which are concerned with loans, capital expenditure and relief of need, he deemed to be sufficiently flexible in character as to permit their contraction and expansion as a partial offset for the rise and fall in the national income" (p. 219).

Support for the active use of deficit financing for stabilisation policy grew further in subsequent decades. However with the stagflation in the Seventies, deficit

Pigou's argument is the traditional tax-smoothing one: if governments determine tax rates on the basis of permanent spending, they can minimise the excess burden of taxation. Public debt would fluctuate in response to the transitory changes in revenue and expenditure. Barro (1979) finds that this model explains public debt dynamics in the USA since 1922. See also Flemming (1988).

finance lost momentum. Critics noted the difficulties concerning the estimate of the actual impact of budget changes on the economy and the risks of fine tuning given the lags between budgetary decisions and their effects on the economy. Theoretical models questioned the possibility for the Government to influence the level of government activity.

In the end, the debate lasting over two centuries has come to justifying government deficits under three set of circumstances: first, when non-remunerative expenses of a wholly abnormal scale have to be financed; second, for financing fixed capital formation; third, when the economy is under unfavourable macroeconomic conditions.⁵ Thus from a strictly public finance perspective the existence of public debt is justified as the cumulated result of deficits incurred under the above mentioned circumstances. These factors should not normally justify large debts: exceptional, cyclical and tax smoothing considerations only justify temporary debt accumulation, while debt for capital formation is somehow limited by the fact that only net investment should be financed by borrowing.

1.2 Monetary policy and financial markets

While public bonds are primarily a tool for financing government, they are also relevant instruments for the development of financial markets and the conduct of monetary policy.

The development of the public bond market can contribute to creating a broad and efficient bond market which can improve the allocation of resources in the economy. In particular, it can help increasing the supply of long-term capital. The introduction of debt instruments which combine high liquidity with low risk can promote household saving (OECD, 1982 and 1993). When public debt is high or grows fast other, more problematic, effects come to the fore (see Section 3).

Government bonds also provide a low risk benchmark for financial markets, thus allowing greater portfolio diversification. This aspect can be particularly relevant for pension funds. If markets are incomplete, government can also increase welfare by introducing new securities that expand risk-sharing (Gale, 1989).

Public bonds can also be used to take care for the needs of specific groups of savers. For instance, government can provide inflation-proof savings instruments which financial institutions may be reluctant to offer (Bach and Musgrave, 1941).

The first two set of circumstances may be especially relevant for local government where the tax bases can be especially small whereas, according to economic theory, stabilisation policy should be mainly managed by the central government. For a more detailed analysis of the role played by debt at the local level see Ter-Minassian (1997) and Ter-Minassian and Craig (1997).

The role and implications of price-indexed bonds represent one of the most debated issue in debt management. This paper does not consider further the issue. See, for instance, Bohn (1988) and Calvo and Guidotti (1989).

Government bonds can also improve the effectiveness of the transmission mechanism of monetary policy. The presence of a low risk borrower issuing significant amounts of bonds at different maturities facilitate monetary authorities in regulating liquidity via open market operations.

In most countries public bonds are the main assets used in these operations. Lacking public bonds the monetary authorities would have to buy and sell private sector assets. This would imply decisions concerning the allocation of resources in the economy. In such a situation there might be pressures towards limiting the independence of monetary authorities (Gokhale, 2002).

1.3 The political economy of public debt

Economic theory points to some rationales for public debt, but both theory and experience suggest that debt accumulation also reflects political factors and that these can push the debt above prudent levels. This Section overviews the debate on the political economy of debt.

The early views

Taking at face value the exceptions allowed to the principle of the balanced budget one would think that the political economy of public debt was not a concern of the classical debate. That would be wrong: as we have seen in the introduction, Ricardo himself feared that debt would induce the citizen to delude "himself with the belief, that he is as rich as before". Much later Puviani (1903) noted that politicians may prefer borrowing to extraordinary levies because citizens underestimate future interest burdens. In 1917 Pareto noted that "Public debt is one way of making acceptable for the citizens what they would not accept with the tax, and this effects is so great that any other considerations of 'burden' is secondary and negligible". Einaudi (1948) noted that while in general borrowing is to be preferred to taxation in funding extraordinary expenditures, this conclusion only holds if the expenditure level is held constant. Since borrowing can lead to greater spending, he concludes that "while from a purely economic and financial point of view, debt is to be preferred to taxation, from a political point of view, that is considering the political decision concerning spending, taxation is to be preferred to debt" (italics in the original text, p. 347).

Indeed the economists who elaborated the exceptions to the balanced budget principle were very much aware of the margins for opportunistic behaviour that they were opening.

Considering extraordinary finance United Nations (1951, p. 61) noted that margins for moral hazard and opportunistic behaviour arise as "the distinction

Pareto (1943), p. 135, as traslated in Chiancone (1986).

between 'ordinary' and 'extraordinary' receipts and expenditure is admittedly not clear-cut, depending ultimately on the judgement of the classifying authority as to whether the receipts and expenditure in question are to continue indefinitely in the future". De Viti De Marco (1953, p. 390, our translation) pointed out that "this subjective element does not allow to define a rigorous and objective rule that draws the line ... between ordinary and extraordinary finance".

Also the distinction between current and capital items retains a certain degree of ambiguity which can be used opportunistically. "The classification procedures which are to be followed in separating "current" and "capital" transactions are among the most controversial and difficult questions in budgetary procedure, especially in view of the frequent abuses of so-called "capital budgets" in hiding deficits which otherwise would have become apparent" (United Nations, 1951, p. 11).

This scepticism was well grounded in experience. "In the case of France, the extraordinary budget was proverbially the dumping place for all expenditures which could not be balanced by tax receipts" (Hansen, 1941, p. 199). In 1945 Keynes notes that in the United Kingdom "the present criterion leads to meaningless anomalies. A new G.P.O. is charged 'below', a new Somerset House 'above'. A Capital contribution to school buildings is 'above' in the Exchequer Accounts and is paid for out of Revenues, and is 'below' in the Local Authority Accounts and is paid for out of loans. The cost of a road is 'above', of a railway 'below'. And so on". "In Canada, although not always realised even by Canadians, a budgetary distinction between ordinary and capital expenditures has been made ever since the confederation in 1867. The official reports show surpluses in fifty of the sixty-six years following 1867; but if the accounting were made on the United States basis, surpluses would appear in only fifteen of the sixty-six years" (Hansen, 1941, p. 199).

Concerning functional finance, the way in which it was first met is exemplified in the following passage: "If I were to pretend I could lay out a programme under which what I borrowed this year would be met by a surplus at the end of three years, everyone would soon perceive that I was only resorting to the rather transparent device of making an unbalanced budget look respectable". 9

All in all it can be argued that the relatively more recent strand of the literature analysing the politico-institutional determinants of government's (possibly excessive) reliance on borrowing is well grounded in the tradition of public finance analysis.

⁸ Quoted in Clarke (1998), p. 65.

⁹ Chamberlain, quoted in Sabine (1970), pp. 15-16.

The recent literature

Recent political economy models of debt can be grouped in six types of models based on:¹⁰

- (i) opportunistic politicians and naive voters with fiscal illusion (voters favour public spending but do not want to pay for it; they do not understand the government's intertemporal budget constraint, politicians take advantage of this and run deficits in order to win elections);¹¹
- (ii) intergenerational redistribution (debt is a way of transferring resources across generations; current generations may exploit the fact that future generations do not vote when the decision on issuing debt is taken);¹²
- (iii) strategic use of debt by the current government (policymakers overissue debt so as to tie the hands of a possible successor with different preferences, in terms for instance of the level or the composition of public spending);¹³
- (iv) coalition governments (debt accumulation may be the result of delayed adjustment to an exogenous shock to the economy as a war of attrition takes place between parties which prefer to shift the burden of adjustment to supporters of other members of the coalition; the initial deficit itself may be endogenised in a model where parties with different ideological motivations coexist in a coalition government; this can be so even if all parties share a preference for a balanced budget);¹⁴
- (v) competition from dispersed interests (if spending decisions are decentralised and revenues are centralised but residually determined there may be a tendency not only to overspend but also to overborrow);¹⁵
- (vi) budgetary institutions (largely based on the idea that deficit bias is the result of a common-pool problem, a large empirical literature compares alternative budgetary institutions, both quantitative limits on fiscal variables and procedural rules and across political systems, and suggests that some institutions are correlated with smaller deficits). ¹⁶

This literature offers some insight as to why countries in similar economic conditions may pursue different debt policies. It confirms the intuition of the earlier studies concerning the risk that myopic or opportunistic behaviour by politicians can result in unsustainable debt dynamics. It also points to the need to introduce rules and procedures restraining budgetary decisions (this aspect is examined in Section 4). However, historical experience shows that the role of these factors can

¹⁰ See the reviews in Alesina and Perotti (1995), Drazen (2000) and Persson and Tabellini (2000).

See Buchanan and Wagner (1977) and the papers in Buchanan, Rowley and Tollison (1986).

¹² See Browning (1975), Tabellini (1990, 1991), Cuckierman and Meltzer (1989).

¹³ See Persson and Svensson (1989), Alesina and Tabellini (1990) and Tabellini and Alesina (1990).

See Alesina and Drazen (1991) and Balassone and Giordano (2001).

¹⁵ Velasco (1999).

For European countries, see Von Hagen (1992) and Von Hagen and Harden (1994). For the USA see Alesina and Bayoumi (1996), Poterba (1994), Bohn and Inman (1996).

change across countries and time and that fiscal developments can be determined by many other factors (such as political ideologies, constitutional constraints and exogenous shocks).¹⁷

Deficit bias and cyclical asymmetry in fiscal policy

Evidence of a deficit bias, likely to be motivated by electoral motives or by myopia, is also provided by a different strand of the literature which investigates the actual reaction of public budgets to changes in macroeconomic conditions. According to European Commission (2001), between 1970 and 2000 "... [in the European Union (EU)] deficits did not fall during periods of high economic growth, implying that countries offset the working of the automatic stabilisers via discretionary tax cuts or, more frequently, expenditure increases; such fiscal relaxation in good times in turn necessitated a tightening during economic downturns" [p. 63]. If discretionary tightening in bad times exactly matches discretionary loosening in good times (*i.e.*, if fiscal policy, though pro-cyclical, reacts symmetrically to the cycle) then this tendency, though negative for the stability of the economic environment, would not imply that fiscal activism *per se* contributes to debt accumulation.

Some evidence of asymmetric behaviour is provided by Buti, Franco and Ongena (1998) for high debt EU countries where, between 1970 and 1990, deficit to GDP ratios are at around 6 per cent of GDP when output is close to or above its trend value while the imbalance increases up to 8 per cent when output falls below its trend level.

Buti and Sapir (1998) also find that in the same period, for the average of EU country, "when there is a moderately negative output gap [...] the actual deficit gradually increases" (even though the reaction to larger negative output gaps is not stronger) while "when there is a moderately positive output gap [...] the actual deficit remains stable" and it is only "when there is a strongly positive output gap [that] the actual deficit improves" (p. 87-88).

Balassone and Francese (2004) test for the presence of asymmetry in the conduct of fiscal policy over the cycle in a sample of 16 OECD countries by estimating the output elasticity of public budgets separately for expansions and contractions. They find evidence that fiscal policy reacts asymmetrically to cyclical conditions as a downturn is usually accompanied by a deterioration of the budget

See Chiancone (1993), who provides a critical review of the political economy studies.

¹⁸ See also Buti, Franco and Ongena (1997).

This evidence is not uncontroversial. Melitz (2002) finds that "...[in EU countries] fiscal policy responded in a stabilising manner in all phases of the cycle but only mildly so" and points out that "...under expansion, the divergence [with Buti and Sapir, 1998] is important". Melitz (2002) also concludes that "...the explosion of debt/output ratios in the EU, and the OECD as a whole, must be explained independently of the cycle" (p. 235).

balance while an upturn does not entail an improvement of the balance. They also find that this asymmetry has significantly contributed to debt accumulation.²⁰

2. The effects of public debt

The issue of the effects of public debt has been at the core of the fiscal policy debate over the last two centuries.²¹ While the issue has been approached in different ways, the debate shows some recurring features.

Debt neutrality and debt burden

The debate on the effects of the public debt goes back to the discussion on debt neutrality (*i.e.*, the equivalence of deficit and tax finance with respect to capital accumulation) and on the intergenerational distribution of debt burden.

Ricardo pioneered the field. He stressed that public borrowing reduces saving and this was for a long time the prevailing view. Against debt neutrality he argued that, due to fiscal illusion (*i.e.*, the inability of agents to correctly anticipate future taxes needed to finance the debt), debt induces a smaller reduction of consumption than taxes do; hence the former exerts a comparatively negative effect on capital accumulation.²² Concerning the generational distribution of the burden of the debt Ricardo argued that the cost of debt is borne when resources are used.

Ricardo's position was taken up in the Forties by the so called "real resources view". According to the supporters of this view, the burden of the debt is borne by current generations as they pay the opportunity cost of financing it. The future servicing and repayment of the debt will only entail transfers from the general tax-payers to the bond-holders ("we owe it to ourselves"). As long as the debt is internally held, these transfers do not alter the overall volume of resources available.²³

The average the debt to GDP ratio in their sample grows by almost 34 percentage points over the period considered (1977-2000), they estimate that almost a third of the increase is due to asymmetric budgetary behaviour.

While this papers focuses on public debt, it is worth recalling that several studies evaluated the issue of the limits and implications of overall indebtedness (public and private). See Summers (1986) for an example concerning the US case.

Between the end of the XIX and the beginning of the XX century, the "Italian school" of public finance explored further the conditions needed for debt neutrality to hold: Pantaleoni and Borgatta focused on the role of bequests; De Viti De Marco on financial markets imperfections; Griziotti on agents' time horizons; Puviani on bounded rationality and fiscal illusion. See, for example, De Viti de Marco (1953), Griziotti (1917) and Puviani (1903). See also the survey by Chiancone (1986). The neutrality hypothesis came to the fore again much later with Barro's (1974) contribution who put it into a fully formalised framework and highlighted the relevance of intergenerational altruism as an alternative hypothesis to the one concerning infinite time horizons.

²³ See Lerner (1943) and Chase (1943).

It was however widely recognised debt finance can affect the rate of capital accumulation and that if these effects are negative, future generations may be said to be burdened by the debt in the sense that they receive a lower capital stock.²⁴ It was also noted that the taxation required to finance interest spending introduces distortions in economic behaviour and involves a deadweight loss (Musgrave and Musgrave, 1984).

In the late Fifties the so-called "utility view" expressed new arguments against deficit finance.²⁵ The focus of the analysis was shifted from social to individual costs. According to this view, the burden of the debt falls onto future generations independently of the effects of debt on capital accumulation. Bond-holders have voluntarily given up resources now to have them back in the future so that their utility is not reduced; future tax-payers, on the other hand, will be forced to transfer (via the tax bill) resources to the bond-holders so that their utility will be reduced.²⁶

Debt and macroeconomic analysis

The issue came to the fore from a different perspective in the Forties and Fifties in the context of the debate on the real balance effect. In commenting on the Haberler-Pigou proposition that changes in the real value of money balances can assure full employment by equating savings and investment at a positive interest rate, Patinkin (1948) noted that the base for wealth effects include the government debt held outside the treasury and the central bank. Tobin (1952) questioned this approach and argued that the additional taxes necessary to finance interest charges may reduce the value of private wealth.

The conventional view developed in the following years, in the context of the neoclassical synthesis, highlighted the long term effects of public debt (Modigliani, 1961). While in the short term deficits affect aggregate demand, in the long term they reduce savings, increase interest rates and reduce productive public capital formation (Diamond, 1965).

The Ricardian equivalence theory contrasted these views and pointed to the macroeconomic irrelevance of the debt/tax mix (Barro, 1974). The theory is based on the consideration that debt implies future taxes with a present value equal to the value of debt. Rational agents proceed as if the debt does not exist. This results in the debt having no effects on the economy.²⁷

See Pigou (1929). Also the supporters of the "real resource view" generally accepted this point.

²⁵ Buchanan (1958) labelled the view that the debt burden is borne immediately as the "new orthodoxy".

See Buchanan (1958), who labelled the view that the debt burden is borne immediately as the "new orthodoxy" and Bowen et al. (1960).

Feldstein (2004) notes that further support to this thesis was provided in the Sixties by the new neoclassical theory of economic growth, arguing that the national rate of saving does not affect long term growth.

The debate that developed in the following years showed that Ricardian equivalence holds only if several conditions are realised. Buchanan (1976) claimed that fiscal illusion precludes the equivalence of taxation and debt. If future liabilities are not recognised, people behave as if bonds are a net addition to their wealth and increase consumption. Moreover, if individuals have finite horizons, they are not altruistic, they are liquidity constrained and they are uncertain about the future, the equivalence does not hold.²⁸

Empirical studies

Several studies tried to evaluate the empirical foundation of Ricardian equivalence. Buiter and Tobin (1980) concluded that the case for debt neutrality is not well eatablished. Feldstein (1976 and 1998), while noting that the promise to pay social security benefits is equivalent to issuing bonds, argued that the empirical evidence concerning the effects of social security wealth suggests that the burden of the debt is shifted via lower saving ratios.²⁹

While several studies rejected Ricardian equivalence,³⁰ other supported the prediction that debt and deficit have no effects on relevant economic variables.

Elmendorf and Mankiw (1998) examine the conventional view of the effects of public debt: debt can affect economic activity both in the short and in the long term. In the short term it increases aggregate demand, in the long term it reduces savings, increases interest rates and reduces productive public capital formation. Ball and Mankiw (1995) note that deficits also tend to reduce net exports and determine an outflow of assets.³¹

Auerbach and Kotlikoff (1987) provide similar indications within an overlapping generations model: in a closed economy in the long run an increase in public debt crowds out private capital nearly on a one-for-one basis.

Gale and Orzag (2002) survey the empirical work concerning the effects of budget deficits in the US and note that the studies that consider deficit expectations in addition to current deficits usually find significant connections between deficits and interest rates. They also note that, since an offsetting increase in private savings is unlikely, "long-term budget deficits reduce national savings and impose substantial costs to the economy, regardless of whether interest rates are affected. ... The reduction in future income is the true cost of a failure of long-term fiscal discipline." (pp. 27-28).

Buchanan and Roback (1987) note that, even assuming that there is no fiscal illusion, there are no effects on consumption and saving only in the very unlikely case in which the fiscal operation involving the debt issue precisely balances off assets and liabilities in each individual's account.

²⁹ See also the extensive survey in Seater (1993).

³⁰ Holcombe *et al.* (1981).

They estimate that the existence of the US debt reduces GNP by 3 to 6 per cent.

Tanzi and Chalk (2000) note that the availability of an income-generating risk-free asset may reduce the resources available to more risky investment options. This may raise the cost of capital for the private sector. They find that in the EU, contrary to Ricardian equivalence, higher debt puts upward pressure on interest rates and is negatively correlated with private investment. Moreover, higher debt is associated with higher tax levels and lower capital spending.

Distortions

High debts may cause distortions in the economy and in public budgets.

The increase in tax burden deriving from higher debt levels is problematic if lump-sum taxation is not feasible and distortions and dead-weight losses cannot be avoided.³² Tanzi and Chalk (2000) note that high debt countries have frequently introduced either regulations or special tax regimes aimed at channelling resources to public bonds. Some regulations have forced financial institutions or other agents to buy public bonds. These solutions reduce the interest burden on public bonds but introduce distortions in the capital market.³³

Government investment spending may be especially affected by the need to keep the deficit under control arising in situations of high and rising public debt. This situation is conceptually equivalent to the introduction of a deficit ceiling and it can be shown that this implies a reduction in investment spending (Balassone and Franco, 2000b).

This may reflect three mechanisms. A first channel works through political economy considerations. Since investment produce deferred benefits policy makers, caring about economic performance only when in power, will invest if they can run deficits and avoid taxing the private sector, while they will avoid investment if its financing must come from taxes in order to keep the deficit under control.

A second channel may reflect a welfare maximisation strategy. Given an expenditure profile, the efficiency loss caused by distortionary taxation is minimised if the tax rate is constant (Barro, 1979). If a deficit ceiling is introduced indivisible investment projects will not allow tax smoothing and may be foregone in order to avoid the ensuing welfare costs.

Finally, since investments produce deferred benefits, the means of financing them (tax rather than debt) also affect inter-generational equity. Tax financing of

Elmendorf and Mankiw (1998) estimate that in the USA the dead-weight loss from servicing the debt is about 0.5 per cent of GDP.

Aiyagari and McGrattan (1998) develop a model in which the optimum quantity of debt depends on the trade off between the benefits stemming from its role in smoothing out consumption and its costs in terms of crowding out capital and requiring distortionary taxation.

investment implies a welfare loss for the current generation and favours future ones: the former fully pays for a project whose benefits will partly accrue to the latter.³⁴

While the results of the literature concerning the returns of public investment are not unambiguous, one can nevertheless argue that the decision to reduce capital spending should derive from considerations concerning these returns and not from the restraint imposed by high interest spending.

Constraints to stabilisation policy

High debt levels may discourage the use of the public budget as a stabilising device for the economy due to concerns over the sustainability of debt. According to European Commission (2001), this was the case in the past decades in the high-debt countries of the EU. These countries frequently had no room for their automatic stabilisers to operate.

Moreover, high debt levels may make deficit finance less effective. The effect of a fiscal expansion on consumption and investment may be reduced in the presence of high debt levels since the private sector is more likely to consider as unavoidable a subsequent compensatory action (e.g., a tax increase). Finally, high debt levels may imply that the effects of expansionary fiscal policy are offset by rising interest rates. It is likely that the effects on interest rates of large fiscal expansions, like the one undertaken by the USA since 2002 (about 6.5 percentage points of GDP), can be limited only if the initial debt level is low (it was 58.9 per cent of GDP in the USA at the end of 2001).

The rationale for having a debt target coupled with a medium term deficit objective in a monetary union can indeed be found in the recognition that in such a context national budgetary policies have an important role for macroeconomic stabilisation in the event of asymmetric shocks and across the economic cycle. The same considerations apply to the use of the public budget in the face of other shocks that would require public expenditure of a wholly abnormal scale, such as natural disasters or wars.

The cost of debt reduction

Large public debts have often determined problematic outcomes. A rising debt ratio may, at some stage, determine a confidence effect which leads to a sharp decline in the demand for public bonds and to a rise in risk premia. These risks are

The link between fiscal consolidation and cuts in capital spending is confirmed by the experience of EU and other OECD countries. See Roubini and Sachs (1989), De Haan et al. (1996) and Balassone and Franco (2000b).

greater when the debt is external. At this stage the government can either significantly improve the primary balance or default its obligations.³⁵

While default is certainly a costly option, it must be borne in mind that the distortions mentioned above with reference to stabilisation and to capital spending are even more relevant when policy makers have to implement abruptly large fiscal adjustment packages. Significant welfare losses may also derive from changes in citizens' expectations. History shows that high debt situations can be brought to an end in many ways (Alesina, 1988). Each process has different redistributive impacts across economic agents and across generations.

3. Debt and sustainability

While there is a theoretical rationale for both allowing recourse to debt and to fear its excessive accumulation, there is no theoretical indication for a specific limit to debt accumulation.

Several studies have approached the issue of debt sustainability considering a specific question: are there limits to debt accumulation in the sense that it should be lower than a threshold above which its burden (as measured by the implied tax rate) significantly affects capital accumulation and growth?³⁶ The answer requires the specification of the equation governing the dynamics of the debt to GDP ratio as a function of budgetary policy (tax, interest and primary expenditure ratios) and of its effects on macro parameters as the rate of interest and the rate of growth.

Unfortunately there is no agreed upon theory of the interactions between the public budget and the economy. The only choice is to use a partial equilibrium framework, assuming that both the interest rate and the growth rate are exogenous to fiscal policy. The partial equilibrium nature of the exercise implies that the possible effects of growing debt on interest rates and growth are overlooked.

This type of analysis was pioneered by Domar (1944) to answer concerns that "... continuous government borrowing results in an ever rising public debt, the servicing of which will require higher and higher taxes; and that the latter will eventually destroy our economy or result in outright repudiation of the debt" (p. 148). Domar showed that a constant overall deficit to GDP ratio ensures convergence of both the debt to GDP ratio and the interest to GDP ratio to finite values. Consequently also taxes needed to service interest payments converge to a finite value as a share of GDP.

The debate on sustainability took a new twist in the Eighties, in connection with the growth of the public sector and the unfavourable demographic trends. It

Default can take different forms. The government can inflate away the debt by raising money supply or devalue the currency (Ball and Mankiw, 1995; IMF, 2003). It can also formally repudiate its liabilities (Eltis, 1998).

³⁶ See the review in Balassone and Franco (2000a).

was spurred by estimates pointing to substantial prospective increases in public expenditure. The development of large welfare systems implied large scale implicit liabilities whose amount is related to the age structure of the population. The additional tax burden required to finance expected expenditure increases became the primary concern. The cost of implicit liabilities would in several countries dwarf that of outstanding public debt. In evaluating public finance sustainability it was no longer sufficient to examine the tax rate implications of a constant deficit, à la Domar. It became necessary to estimate the future deficit path implied by current policies.

However, from a formal point of view, Domar's definition of sustainability was still acceptable. Blanchard *et al.* (1990) proposed two necessary conditions for sustainability: (a) "... that the ratio of debt to GNP eventually converges back to its initial level ..." (p. 11); (b) that "... the present discounted value of the ratio of primary deficits to GNP ... is equal to the negative of the current level of debt to GNP ..." (p. 12).³⁷

With respect to the necessary condition for sustainability used in Domar's paper (convergence of the undiscounted debt ratio to a finite value), the first definition in Blanchard *et al.* is tighter; but it is so at a cost of arbitrariness. Domar's model cannot specify the maximum sustainable debt level and Blanchard *et al.* do acknowledge that "... the justification for the ratio to eventually return to its initial level, as opposed say to zero, or to a higher but stable level, is, however, much less evident ..." (p. 11).

On the contrary, the second definition is looser than Domar's one: an ever-growing undiscounted debt ratio is allowed. As the authors explain, "... this is because of discounting, which implies that things far in the future do not matter much for today ..." (p. 14). However, as pointed out by Artis and Marcellino (1998), "... this suggests that both quantities should be analysed and not only the discounted one ..." (p. 6).

All three conditions have been employed in empirical studies on sustainability so that it is not always clear that different authors are talking about the same thing when they try to assess the "sustainability" of public finances.³⁸ The absence of a clear-cut theoretical benchmark to assess sustainability has often

Blanchard et al. (1990) treat the two conditions as equivalent (pp. 11-12). However, as the authors acknowledge (p. 14), there is a difference while the first condition implies the second, the latter is necessary but insufficient for the first to apply.

Several studies tested the sustainability of past budgetary policies and shed light on the soundness of the hypothesis of Ricardian equivalence in macroeconomic modelling. Aschauer (1985) and Seater and Mariano (1985) tested the hypothesis that governments' receipts must equal expenditures in present-value terms jointly with a permanent income hypothesis. Hamilton and Flavin (1986) were probably first in testing the present value budget constraint per se.

favoured the use of *ad hoc* definitions.³⁹ The problem also affected the definition of European Monetary Union's (EMU) fiscal rules.

The sustainable level of public debt varies between countries depending on several economic and political features (IMF, 2003). It is usually higher for industrialised countries than for emerging market economies. The former have historically shown the capability to generate large enough primary surpluses to ensure the sustainability of their debt even in adverse circumstances. The latter have generally not gained this credibility. This is due to, for example, to weak revenue bases (with lower yields and higher volatility) and to less effective expenditure control during economic upswings (this is particularly the case in Latin America). However, among emerging economies there are significant regional differences, with Asian countries generally doing more to ensure debt sustainability than countries in other regions.

4. Control mechanisms

The previous Sections have examined the factors that may lead to excessive debt accumulation and the consequences of this accumulation. This Section considers the mechanisms which can control debt expansion. Two issues are considered: can the financial market restrain debt growth? If rules are deemed necessary, which rules should be adopted?

4.1 Market

Default premia and credit constraints clearly have the potential for disciplining irresponsible sovereign borrowers. Default risk is priced by the market. Thus market-based fiscal discipline would initially take the form of a rising default premium on the debt of a country running excessive deficits. If these deficits persist, the default premium would increase at an increasing rate until, eventually, the offending country would be denied access to additional credit. The increase in the cost of borrowing, along with the threat of reduced availability of credit would then provide the incentive to correct irresponsible fiscal behaviour. The key question is whether and under what conditions credit markets restrain irresponsible borrowing and thus irresponsible debt accumulation.

Lane (1993) identifies the following necessary conditions to be met for the market to be effective: a) no government unit should have privileged access to the market; b) the market must have access to all the information necessary to evaluate the financial reliability of each unit; c) the bailing-out of troubled government units must not be allowed.

We are leaving aside the issue of external debt sustainability. This is especially relevant for less developed countries and emerging market countries (see Chalk and Hemming, 2000 and IMF, 2003 and the references therein).

If any of the three conditions above is not met, market signals risk too late and the change in market perception of the state of government finances may change too suddenly with possibly disruptive consequences.

The possibility to rely on market-induced fiscal discipline was considered when defining the conditions to be met by a country for joining EMU. At that time it was pointed out that the expectations of a partial bail-out would be sufficient to make the fiscal stance of governments insufficiently reflected in credit risk premia (e.g., Lamfalussy, 1989, and European Commission, 1990). Moreover, accounting practices for general government accounts – even in national accounts – provided further obstacles to effective risk assessment by market agencies and investors. Finally, it was argued that the international evidence on sovereign defaults suggested that "constraints imposed by market forces might either be too slow and weak or too sudden and disruptive" (Committee for the Study of Economic and Monetary Union, 1989, 24). In the end, regulation was seen as a necessary supplement to market forces.

The crisis in which the Stability and Growth Pact has plunged in 2003 may induce to reconsider the issue today. As to the three conditions laid down by Lane, it is fair to say that government's privileged access to the market was never an issue (the European Central Bank is independent, and financing of governments is no longer possible), but the other two conditions remain problematic. Some problems remain open with reference to the information available to market agents on government finance. Nevertheless, this information has greatly improved thanks to the statistical requirements provided for by the Treaty and the Stability and Growth Pact and the consequent markedly increased homogeneity of available accounts. Also the credibility of the no-bail-out commitment still remains an open issue at least for large highly indebted countries playing a major role even in the large European financial market.

Recent empirical work confirms the conclusions concerning the role of market incentives reached at the end of the Eighties. Rating can help but also seems to react slowly (Balassone, Franco and Giordano, 2004). In this regards, FitchRatings (2004) observes that it is unlikely that financial markets can give a strong incentive for fiscal discipline to euro-area sovereign borrowers, since "a euro-area government whose budgetary position weakens is likely to have to pay more for its debt, but that extra cost will be small."

4.2 Rules

Regulation may support the market in providing proper incentives for fiscal discipline. Indeed rules are used in most countries. 40 Budgetary institutions may be partitioned in two broad categories: legislated quantitative limits on fiscal variables and procedural rules.

⁴⁰ See Kopits and Symansky (1998) and the papers in Banca d'Italia (2001).

In order to restrain debt growth, one may consider either rules directly setting a limit to debt levels or rules controlling debt levels via limits to deficit levels.⁴¹

In practise, quantitative limits to the stock of debt are rare: budgetary institutions usually aim at controlling the deficit. This is so essentially because governments can take decisions concerning the yearly deficit while the stock of debt reflects the budgetary decisions taken in past years. Moreover, a ceiling on debt would only constrain policy makers once the debt level comes close to it; at that stage the needed correction may be too large to be politically viable and the ceiling itself may have no credibility. By the same token, the credibility of a debt target set when the current stock is already too high hinges upon a sustainable path of fiscal consolidation. Therefore either there are no explicit limits to the debt level or when quantitative limits to debt are used, they are usually accompanied by deficit rules.

Lower levels of government are frequently subject only to deficit restrictions. In some countries, fiscal targets are specified by the law; in others, they are the outcome of budgetary procedures in which both cooperation and controls may be present. Administrative controls can also be used. Balanced budget requirement are frequently applied. In federal countries and in countries characterised by a high degree of decentralisation, recourse to debt is generally permitted to any government tier. The rules generally limit the overall size of the deficit (either directly, or indirectly via thresholds for interest outlays) and allow indebtedness for certain purposes only (usually public investment). The constraint on indebtedness generally applies *ex ante*: possible overshoots may be compensated for in subsequent financial years. Further budgetary flexibility is sometimes provided by the so-called rainy-day funds. In the latter case, gross and net debt positions can significantly differ.

These restrictions tend to be less common for sovereign governments. However, where there are no numerical deficit limits, the nature of the budgetary process (*i.e.*, the system of rules and regulations, both formal and informal, which determine fiscal decisions) is extremely important in determining the size of deficits. Empirical research has indicated that deficit levels tend to depend on the power of the Treasury Minister, on the government's ability to enact supplementary budgets during the fiscal year, on the power of the executive to enforce the original budget and on the role played by amendments on budgetary procedures.⁴⁴

Spaventa (1987) considers whether a fiscal rule respecting the government's intertemporal budget constraint is sufficient for ensuring the sustainability of fiscal policy. He concludes that, since the debt ratio can rise to very high levels, the constraint is not sufficient when the size of the debt can affect the interest rate or there is a limit to the tax burden.

⁴² See Ter-Minassian and Craig (1997), Banca d'Italia (2001) and Balassone, Franco and Zotteri (2004).

For example, this is the case in the USA. For a detailed analysis, see Knight and Levinson (1999) and McGranahan (1999).

See von Hagen and Harden (1995) and Ferejohn and Krehbiel (1987). From an empirical perspective, von Hagen (1992) builds up an index of the institutional characteristics of the budgetary process in EU countries which accounts, *inter alia*, for the strength of the prime ministers in budget negotiations, limits to amendments, budget voting process, the degree of transparency. He finds that, in the Eighties, countries with an high index value were those with lower deficit and debt levels.

Other countries adopt a golden rule approach. In Germany, for instance, yearly deficits are allowed up to the level of gross investment in the federal budget. ⁴⁵ In the UK, legislation providing for the net public debt as a proportion of GDP to be held at a stable and prudent level over the economic cycle (this level is currently set at 40 per cent) is accompanied by a golden rule requiring that, over the economic cycle, the government borrow only to invest. ⁴⁶

In the USA a statutory limit on federal government gross debt is in place since 1917.⁴⁷ However Buchanan and Wagner (1967) note that the legal status of the debt limit is not clear, since the excess of appropriations over revenue collections might require debt creation in conflict with debt legislation. Over recent years this limit has been supplemented by several rules concerning the budget balance and public spending (Peach, 2001). After the failure of the 1985 Gramm-Rudman-Holling Deficit Reduction Act to attain the set budgetary targets,⁴⁸ the possibility of introducing a balance budget principle in the Constitution was discussed.

EMU provides an interesting case of fiscal rules aimed at controlling both debt and deficit levels and of the ensuing problems. 49 The Treaty of Maastricht in 1992 stated that gross debt must be below 60 per cent of GDP or, if above that level, it must be decreasing at a satisfactory pace and that the deficit must not exceed 3 per cent of GDP, unless exceptional circumstances apply. In 1997, with the Stability and Growth Pact, budgetary flexibility was explicitly pointed out as an objective to be achieved along with fiscal soundness. The Pact spelled out the circumstances that may allow a deficit above 3 per cent. At the same time it introduced a medium term target of a position close to balance or in surplus, which is now interpreted in cyclically adjusted terms. In this way the Pact both tightened the Treaty's deficit rule and tried to reconcile it with the possibility of counter-cyclical fiscal policy. However the Pact left the Treaty's rule on debt somewhat at the back of the picture. It did not specify how the requirement of a satisfactory pace of reduction was to be interpreted in practice.⁵⁰ As a result the focus of the debate concerning the assessment of public finance conditions in EMU countries was strongly biased towards the deficit.

Art. 115 of the Constitution states that: "Borrowing cannot exceed the total investment expenditure in the budget; exceptions are only allowed to avoid disturbances to the overall economic equilibrium". The items mentioned partly differ from those considered in national accounts.

⁴⁶ See HM Treasury (1998a) and Kilpatrick (2001).

⁴⁷ See Peach (2001).

⁴⁸ See Gramlich (1995)

⁴⁹ The economic policy framework of EMU is extensively examined in Buti and Sapir (1998), Buti, Franco and Ongena (1997) and Brunila, Buti and Franco (2001).

Indeed the requirement to keep the debt ratio declining could prove to be inconsistent with the attempt at providing budgetary flexibility along the cycle (Balassone and Monacelli, 1999). This concern is perhaps one of the rationales for the requirement that the debt stays below 40 per cent in the UK's fiscal framework to apply only over the length of the economic cycle rather than on a annual basis and to develop a cyclically adjusted measure of debt (HM Treasury, 2002).

More recently, in the context of a debate on the rationale and the effectiveness of EMU's fiscal framework, several proposals have been put forward to give more weight to debt. Recently the ECOFIN (2001), the Economic Policy Committee (2003) and the European Commission (2003) have reaffirmed the importance of debt in fiscal surveillance by stating that the assessment of countries' public finances must take account of the need for a rapid decline in debt among those countries which have high debts and that greater weight must be attached to debt over future years. Several specific proposals to differentiate the rules applying to countries on the basis of their debt levels and medium-long term fiscal prospects have been put forward. While greater reference to the debt ratio does not raise measurement problems, reference to implicit liabilities is more problematic: long-term estimates are subject to considerable uncertainty related to the macroeconomic, demographic and behavioural scenarios (see Section 5.1).

The twin track approach of setting both a deficit and a debt constraints seems reasonable also in view of implementation problems (see IMF, 2001).

First, deficit controls are effective in restraining debt dynamics only if the reference deficit measure is a comprehensive one (see Section 5). A maximum debt reference level can therefore be a useful companion of deficit-based rules because it compensates for possible shortcomings of the deficit measure. Moreover, if deficit rules are defined as ceilings, a prudent debt reference level can help avoid the deficit is kept constantly close to the ceiling.

Second, although targets for the debt ratio may install incentives for governments in the long term, there may be some room for manipulation, via, for example, sale-and-lease-back operations. Moreover, it cannot be taken for granted that deficits do not matter if debt levels are under control. In the context of EMU, for example, high deficits may negatively affect the possibility to attain a good policy mix.

5. Measuring public liabilities

The effectiveness of rules crucially depends on the indicators they refer to.⁵³ Not only has the indicator to be relevant in view of the purpose of the rule, it also has to be based on a transparent and unambiguous operational and statistical framework. Experience highlights the possibility that policy-relevant indicators can be manipulated. It also points to the dangers inherent in statistical frameworks allowing for different interpretations of accounting rules.⁵⁴

See Buti, Eijffinger and Franco (2003) for a critical review of the debate.

⁵² See Buti et al. (2003), Pisani-Ferry (2002), Wren-Lewis (2003), Wyplosz (2003), HM Treasury (2004).

See Balassone, Franco and Zotteri (2002).

⁵⁴ See Balassone and Franco (2001).

In particular, public debt can be defined in many different ways, depending on the sector of reference and the liabilities to be considered. As in the case of other fiscal indicators, the choice depends on the policy purpose of the indicator as well as on operational considerations. This Section examines the main methodological aspects concerning the measurement of public liabilities in general (Section 5.1) and with specific reference to the EMU context (Section 5.2).

5.1 From financial debt to overall fiscal liabilities

First of all, the sector of reference should be defined. In principle, the definition should include all the public bodies whose financial behaviour has an impact on government finance and ultimately on taxpayers. All levels of governments and other public bodies, such as social protection institutes, should be considered. In practice, there is a grey area, which regards, in particular, public enterprises (Levin, 1993; Stella, 1993).

Second, the relevant liabilities should be defined. One can choose between gross and net measures, face values and market prices, financial debt and a broader definition including non-financial debt. The latter can include current non-financial liabilities (e.g., taxpayers' credits, public guarantees, etc.) and also perspective budget liabilities, such as the pension rights granted by pay-as-you-go (PAYG) schemes. The following subsections examine these issues moving from the more traditional definition of public debt to broader definitions. ⁵⁵

Financial liabilities

Public debt has been traditionally defined as the sum of financial liabilities issued by all government levels.⁵⁶ The assets owned by governments are not usually deducted. This definition reflects practical considerations: financial liabilities can be timely and accurately estimated while the assessment of assets is sometimes problematic.

The gross debt definition overlooks the fact that assets owned by government can be sold to repay the debt. A net debt measure represents a better benchmark for assessing fiscal sustainability. However, this solution raises some difficulties: first, the degree of liquidity of government assets should be taken into account; second, data on assets are often subject to significant uncertainty, especially those on non-interest bearing assets.

These considerations may suggest relying both on a gross and a net debt definition is preferable. The former is more precise, more timely available and more relevant over the short term, the latter is more economically relevant in a longer time perspective.

⁵⁵ See Blejer and Cheasty (1993), Kopits and Craig (1998) and Premchand (1995).

The liabilities issued by a public body and owned by another public body are netted out.

The issue of valuation can be addressed from different points of view. For governments, market valuation is not the relevant measure. It refers to the sum the government would be asked to pay if it were to buy back its debt before it falls due, but the government has no obligation to do so. In evaluating its solvency, therefore, the relevant price is the one to be paid when liabilities fall due. Furthermore, reference to market values makes the debt measure extremely volatile. For investors, the market value of government liabilities matters only in so far as they intend to sell or buy such liabilities on the market. However, in evaluating government solvency, also investors should look at redemption values.

Once the debt measure has been chosen, the deficit should be defined accordingly, *i.e.*, in such a way as to take into account all transactions determining a change in debt. If the debt is gross, the deficit should reflect both non financial transactions and financial transaction in assets.

Government net worth

In recent years some studies have extended the analysis to a broader view of the government balance sheet, considering both public assets and other explicit or implicit liabilities. These developments are closely related to several theoretical studies that have pointed to the deficiencies of conventional cash-flow deficit measures in the assessment of fiscal impact and of budgetary sustainability. As to the latter, it has been suggested that conventional accounting methods do not allow to adequately monitor and control the government's overall fiscal position.

In order to overcome these difficulties it has often been prescribed to resort either to "economic deficit" or to "government net worth". These definitions would consider public assets and all "contingent liabilities". A contingent liability can be defined as a public sector action that determines a cash expenditure only if and when a certain event takes place. Contingent liabilities may emerge either from the government involvement in the economy (guaranteeing the debt of public enterprises or deposit insurance) or from its commitment to provide services or other transfers in the future.

The inclusion of future spending commitments is consistent with the forward looking nature of sustainability analysis, which cannot be based on "point-indicators" which only depict the current budgetary situation (Balassone and

If a net measure of debt is used, symmetry would require that assets be valued in the same way as liabilities. However, an argument for market valuation of assets in any case could be made, based on the consideration that they can only be sold at market prices.

See, respectively, Kotlikoff (1984) and Buiter (1983). For a critique see Mackenzie (1989). For a survey see Towe (1991) and Blejer and Cheasty (1991).

Both these solutions would require, for instance, the inclusion of pensions in fiscal accounts when obligations arise rather than when the actual payment is made. In order to evaluate economic deficit, contributions to public PAYG schemes would have to be classified as a financing item, while pensions would be considered as a loan repayment or as an interest payment. Any change in the present value of pension liabilities would immediately influence government net worth.

Franco, 2000a). The assessment of future developments can refer either to long term projections of public expenditure or to summary indicators of these projections.

From the mid-Eighties an increasing number of studies have examined the long-term prospects for public budgets. ⁶⁰ These studies usually focus on those public expenditure items which are particularly dependent on the population age structure (such as pension, health, education) and seek to assess the likely change of these expenditure on GDP. While several economic, political and social factors can obviously affect the dynamics of per capita transfers and services, the studies examining the prospects of age-related expenditure usually focus only on two rather specific factors: the effects of changes introduced in legislation, but not yet embodied in present expenditure profiles, and of structural expenditure trends. These two factors are considered because they are consistent with a constant policy approach: the estimates define the liabilities embodied in current policies. ⁶¹

Summary indicators of the outcomes of long-term projections have been proposed to "... summarise their results through a single number which comes as a simple metric, allowing for a simple interpretation of the result ..." (Blanchard et al., 1990, p. 32). Buiter (1985) suggests to use the difference between the current primary deficit and the one that would allow a constant net worth to GDP ratio. Blanchard et al. (1990) define the tax gap as the difference between the current tax rate and the one that would guarantee that the debt ratio comes back to its initial level over a specified period. Summary indicators of "... the fiscal burden current generations are placing on future generations..." (Auerbach et al., 1991, p. 55) are provided by generational accounts, the most recent product of a line of research criticising traditional point indicators, based on long-term projections of fiscal variables appearing in the general government present value budget constraint.

In general summary indicators meet the same difficulties as straightforward long-term projections. In addition, it may be noted that they do not provide information about the timing of the effects of demographic changes. Concerning generational accounting, moreover, the interpretation of results is not intuitive. This may hamper their use for policy purposes. 62

⁶⁰ See Franco and Munzi (1996) and Franco and Marino (2004) and the references therein.

The most basic approach provides estimates of the effects of demographic changes on public expenditure under the assumption that age-related per capita expenditure levels remain constant either in real terms or in per capita GDP terms at the initial level over the projection period. In other words, it is assumed that present standards of transfers and services are maintained for all population age-groups and that there is no behavioural response from governments and households to demographic changes.

For a critical assessment of generational accounting, see, e.g., Buiter (1995), Hagemann and John (1995), Haveman (1994) and IMF (1996). The main problem specific to generational accounting is the upward bias its methodology induces in the assessment of the effort needed to ensure solvency due to the different treatment of fiscal variables, to the different rules applying to present and future generations and to the fact that future effects of legislative changes already introduced are not taken into account.

Pension liabilities

The issue of contingent liabilities is particularly relevant for pension schemes. In recent years a number of studies have estimated the liabilities of PAYG pension schemes⁶³ and have argued that these liabilities should be taken into account when evaluating the state and the perspectives of public finances.⁶⁴ It has been claimed that "the strains that higher dependency ratios will impose on budget policies can be seen by examining the present value of future net liabilities of the pension systems in the major industrial countries." (IMF, 1993, p. 56).

Estimates of pension liabilities may represent a useful complement to conventional debt and deficit measures.⁶⁵ However, the ratio of accrued pension liabilities to GDP is not an indicator of pension schemes' sustainability:⁶⁶ a high liabilities to GDP ratio does not necessarily imply an imbalance in PAYG pension schemes; nor does it imply that an imbalance will occur in the future. Any judgement about the sustainability of pension schemes and the pressure they exert on public budgets requires estimates about the resources available to pay for the accrued pensions, namely about the evolution of employment and per capita income.⁶⁷ The sustainability issue should be addressed with other indicators, such as the pension expenditure to GDP ratio and the contribution rate that assures the cash balance of pension schemes.

There are practical as well as theoretical reasons not to include accrued pension liabilities in the deficit and debt statistics used in defining and evaluating fiscal policy. Pension liabilities are uncertain and depend on the specific assumptions adopted upon a variety of factors. Pension rights are not embodied in formal contracts and are not tradable (the debtor can modify both the timing and the amount of the payment even taking individual characteristics into account). ⁶⁸

5.2 Monitoring debt trends: the case of EMU

The reference accounting framework for EMU rules is specified in the Protocol on the excessive deficit procedure. Eurostat oversees the correct

Critical to such estimates is the assumption that revenue cannot be increased proportionately to the expected growth of expenditure. Indeed, the basis for estimating the net present value of the implicit debt measures is the difference between the expected growth of expenditure and revenue under a constant policy scenario (Heller, 1986).

⁶³ See Van den Noord and Herd (1993 and 1994), Kuné et al. (1993), Hagemann and Nicoletti (1989), Hills (1984), Beltrametti (1993) and Rostagno (1995).

⁶⁴ See Van den Noord and Herd (1993), IMF (1993), Hoffmann (1993) and Castellino (1985).

⁶⁵ They provide a measure of the cost of terminating PAYG pension schemes when complying fully with present benefits rules.

⁶⁶ See Franco (1995).

These problems are even more important when referring to other social benefits, such as medical care, education and welfare benefits. It may be debatable to what extent the government is legally obliged to meet such potential obligations and what is the quality and quantity that is to be taken into consideration in the estimates (Heller *et al.*, 2003).

implementation of definitions and computational criteria by national statistical institutes and releases explanatory notes concerning controversial issues.

In this accounting framework, the deficit is identified with net borrowing as defined in the European System of Integrated Economic Accounts (ESA). ⁶⁹ The debt is defined as gross financial liabilities at nominal (face) value consolidated between and within the sectors of general government. Although the debt measure is not defined in ESA, the relevant financial instruments are those specified within that framework.

EMU statistical indicators can be assessed both with respect to the objective of fiscal soundness pursued with fiscal rules⁷⁰ and with respect to the margins for interpretation allowed for by the underlying operational and statistical framework (transparency).

With respect to the objective of fiscal soundness, reference to a gross measure of debt seems in line with sustainability analysis, although net debt would provide an important complementary information (see Section 5.1). Reference to nominal value does not seem appropriate as it does not always coincide with redemption value, which is the relevant measure for assessing sustainability (see Section 5.1). Moreover EMU rules are based on traditional point indicators; the support role to be played by long term projections is not sufficiently specified. ⁷²

As to transparency, while reference to a predetermined and independently defined statistical protocol is a positive trait of EMU's framework, the deficit and debt measures adopted are not mutually consistent (*i.e.*, the ESA95 deficit is not the flow concept corresponding to changes in the stock of gross financial liabilities), which is a clear *vulnus* to transparency.

ESA is a predetermined and independently defined protocol. This represents a significant advantage in terms of transparency (especially in a context where international comparison is crucial); nevertheless no protocol can be detailed enough to specify the intended meaning of its provisions with respect to all possible circumstances. This implies that *ad hoc* decisions have to be reached when countries introduce new accounting operations. At times compromise solutions are introduced. However, the need to adopt solutions with an *erga omnes* validity has contributed to limit the number of *ad hoc* exceptions and adaptations.

EMU fiscal rules were designed to preserve a sound fiscal stance and to allow for budgetary flexibility in the face of adverse circumstances. Here we focus only on the first objective for which debt dynamics matter most. Indeed, the use of both a deficit and a debt measure can be problematic from the point of view of the flexibility target (see Section 4.2).

One example is the valuation of Italian Post Office Deposit Certificates whose nominal (face) value does not include accrued interest which will have to be paid at withdrawal of funds (similar bonds are issued in Portugal). Market valuation (the criterion adopted in ESA95) would not represent a satisfactory solution. Market valuation refers to the sum the government would be asked to pay if it were to buy back its debt before it falls due, but the government has no obligation to do so. Furthermore, reference to market values would make the debt measure extremely volatile.

The need for a forward-looking assessment of the budgetary situation is somehow taken into account in the Excessive Deficit Procedure which foresees the submission by member-states of multi-year programs including medium-long term projections whose internal consistency, underlying assumptions and, ultimately, attainability are subject to scrutiny. Long term projections are receiving increased attention in the monitoring of budgetary trends (section 4.2).

The deficit indicator accounts only for real transactions and is mainly based on accrual accounting, while the debt indicator also accounts for the impact of financial transactions and is based on cash accounting. Moreover, two other factors may create a gap between the deficit and the change in the debt level: (a) exchange rate fluctuations affect the whole stock of foreign currency denominated assets and liabilities (they determine a change in net debt) but are not reflected in the deficit as this is only concerned with actual transactions; (b) different accounting conventions are adopted for recording the effect of transactions on stocks and on flows with respect to liabilities not issued at par and to liabilities denominated in foreign currency.⁷³

The difference between the change in the debt and the deficit measure chosen for EMU rules was by no means negligible over the Nineties; the yearly average for EU countries between 1992 and 2001 was almost 1 per cent of GDP.

This leads to a problem with reference to debt sustainability because of the inconsistency between the deficit and the debt indicators. Indeed, compliance with the deficit rule not only does not guarantee compliance with the debt rule but also it is, in principle, compatible with any debt dynamics. Since the deficit rule is somewhat more important, more binding and more "operational" that that debt rule, the inconsistency problem is best tackled with reference to the deficit rule.

Two changes could be considered, characterised by different degrees of complexity and by different impacts on the existing fiscal framework.⁷⁴

The simpler change would be to refer to cash accounts when compiling the deficit measure presently adopted. This would increase transparency and timeliness of data, thereby improving the effectiveness of both the rules and the monitoring process. While cash data are not immune from window-dressing (e.g., by delaying payments to providers or to employees), it is likely that somebody would voice against such practices (the providers of goods and services and the recipients of salaries and social benefits). This is not the case for opportunistic accrual accounting.

The more complex change would be to redefine the reference deficit as the change in nominal debt. This would have further benefits in terms of timeliness and transparency. As to the former, data on financial liabilities are available more rapidly than data on real transactions and on transactions in financial assets (generally the information set for the general government is complete within a

Concerning the first aspect, the nominal value of liabilities affects the debt while it is the price actually paid by the creditor that corresponds to the deficit. Concerning the second aspect, foreign currency debt is converted in domestic currency values based on end-of-period exchange rates, while the value of the transaction corresponding to the deficit is the one computed on the basis of the exchange rate at the time of the transaction.

⁷⁴ See Balassone, Franco and Zotteri (2002).

month after the end of the reference period). As to the latter, data are usually publicly available from market sources. 75

This change would also imply the use of a comprehensive deficit measure, in line with the focus of sustainability analysis. Moreover, it would increase the consistency of EMU fiscal framework as it would amount to collapsing the two indicators currently used into one. By applying the new deficit definition to both the medium term target of close to balance or in surplus and the 3 per cent threshold, debt dynamics would also be under control.

6. Debt management and fiscal rules

The main objective of debt management is usually to minimise the cost of covering government's borrowing needs, given a certain level of acceptable risk.

This objective is complemented by other objectives which can be split into two groups. First, there are those relating directly to the conduct of government borrowing operations such as ensuring the government's continued access to financial markets, ensuring an effective and efficient liability management with regard to costs and risks, achieving a balanced maturity structure and ensuring an effective management of the new issue operation and an efficient functioning of the secondary market for government securities (Tobin, 1963). Second, there are objectives related to other policies such as "coordination" with respect to monetary policy, improving the functioning of financial markets at large, developing the bond market as a whole, promoting household saving and contributing to a better distribution of income and private wealth (OECD, 1993).⁷⁷

Complementary objectives are likely to play a different role from country to country and their role may as well change over time within the same country (indeed, the role played by complementary targets may depend, for example, on the size of public debt; OECD, 1999).

Some recent developments reflect changes in the relative importance of the objectives of debt management. In particular, since the late Nineties in many OECD

Or, at least, they are available not only from government sources but also from the counterparts of the underlying financial transactions.

The change would imply the inclusion in the deficit measure of the difference between the nominal value of bonds and their price at issuance, which is fine within the context of sustainability analysis. On the contrary, the change in nominal debt ought to be measured net of the effects of exchange rate movements via foreign currency denominated government bonds (which are not under government's control). However, the opportunity and necessity of using this kind of financial instruments may also need to be reassessed.

For example, HM Treasury (1998b) states that "[t]he primary objective of debt management policy is to minimise the long-term cost of meeting the government's financing needs, taking into account risk, while ensuring that debt management policy is consistent with the objectives of monetary policy. The government will meet this objective by: (1) determining the maturity and nature of the government debt portfolio, through managing the maturity and composition of debt issuance; (2) pursuing debt management policies that are open, predictable and transparent; (3) developing a liquid and efficient gilt market; (4) offering retail savings instruments through National Savings which provide cost effective funding.

countries the supply of government bonds has become more concentrated (*i.e.*, government issue large amounts of a small variety of bonds). In the secondary market, debt management has become more active. In both the primary and the secondary markets, improvements depends on the extensive use of electronic tools for issuing and exchanging bonds (Scarpelli, 2002). All these developments can be linked to the target of increasing the liquidity of the government bond market so as to reduce the cost of debt: the more liquid the market is, the lower the premium that investors are likely to require on top of the government bond return.⁷⁸

The introduction of fiscal rules may affect debt management either via a change in the macroeconomic environment or via a change in the priorities of the manager (indeed the cost of non compliance with the rule enters the manager's objective function). Specifically, fiscal rules can affect the choice of the maturity structure and the indexation features. We consider two aspects: the first refer to the short-term potential effects of fiscal rules on debt management, the second regards some possible medium and long-run effects.

Short term issues: smoothing the budget balance and window dressing

If the debt manager has an objective function which explicitly considers the instability of the budget balance as a cost, the optimal debt composition depends on the correlation between output, inflation and interest rates (Lucas and Stokey, 1983). If output and interest rates are negatively correlated, then a long term debt maturity would limit the effects of unfavourable interest rates shock when the debt would anyway tend to worsen (via the negative effect of the cyclical conditions on the budget balance). Moreover, the budgetary effects of shocks to interest rates decrease with the debt maturity. Thus, the optimal debt maturity lengthens with the volatility of interest payments. Obviously, the importance of all these effects increases with the debt size.

First, the introduction of a fiscal rule changes the terms of references used by the debt management to judge the stability of the budget balance. For instance, in the EMU context, the 3 per cent limit on the deficit to GDP ratio has to be accounted for. The debt manager can use maturity and indexation to hedge against inflation and output shocks to the budget so as to stabilise the deficit to GDP ratio below the 3 per cent threshold (see Giordano, 2001 and Missale, 2003). The impact on debt management is even stronger if the debt manager considered budget stabilisation as irrelevant in determining debt costs before the introduction of the rule.

Moreover, given the overall debt cost (in present value terms) that the manager is willing to bear, a fiscal rule can affect the choice concerning the mix of

This point is also stressed by FitchRatings (2004) and by the ECB (2003). These developments are particularly visible in EMU countries. Indeed, since all government securities issued by EMU countries are denominated in euro, the liquidity and the risk of these bonds have become their distinguishing features. EMU participation also contributed to an increase in competition among governments in selling their bonds.

present and future costs. Specifically, the increase in current potential costs due to possible non-compliance introduces incentives in favour of financing tools which shift costs from today to tomorrow (e.g., swap operations).

Finally, the introduction of a rule can induce the debt manager to use financing tools which are more costly, but are not classified as public debt. These include securitisation operations and the building up of financial networks ensuring that debt is issued by entities outside the general government rather than by the general government itself.⁷⁹

With reference to the US statutory limits to federal gross indebtedness, Buchanan and Wagner (1967) stress that "in order to avoid the necessity to ask for continuous shifts upwards in the legal limits, government agencies and officials attempt to finance various federal expenditures through ways that do not fall within the nominally measures totals of public debt".

Long term issues: declining debt ratios

Fiscal rules generally aim at creating a low-deficit environment. The Stability and Growth Pact specifies that each country should aim for a medium term objective of a budgetary position "close to balance or in surplus". This would set the debt ratios on a declining trend (if the impact of stock-flow adjustment does not offset that of the budget balance). Asymptotically, debt ratios would converge to zero, to negative ratios or to relatively low levels, depending on the target set and the stock-flow adjustments.

This drives the attention to the policy options available to policy makers in this new environment characterised by low and declining debt:⁸¹ the key challenge would be to minimise the negative side effects of the projected decline in debt. Indeed, as already mentioned, governments' bonds have an important hedging and pricing role, they guarantee the liquidity of the bond market and make monetary policy signals more transparent.

OECD (1999) suggests two reasons for maintaining a minimum level of gross debt while possibly reducing net debt.⁸² First, private debt bonds are not perfect

In recent years some European countries tried to reduce their general government debt-to-GDP ratio by moving some units (and therefore their debt position) from the general government sector to outside this sector

According to the guidelines of the European Council, compliance with the Stability and Growth Pact should be assessed considering the cyclical position of the economy. Even though not explicitly phrased in these terms, EMU rules may therefore be interpreted as requiring that each member state choose a budgetary target in cyclically adjusted terms and let automatic stabilisers or discretionary action operate symmetrically around it without bringing the nominal deficit above the 3 per cent limit.

⁸¹ Gokhale (2002) considers these issues with reference to the US experience. See also IMF (2001) and Bohn (2002).

This could be done by investing government surpluses in private financial assets. This would drive debt management to face a new problem: how to build up an optimal position in private-sector assets. In particular, investing in private assets raises the issue of dead-weight losses: this type of investment is (continues)

substitutes for public debt bonds (see Section 1.2). Second, once gross debt is fully repaid, should public budgets turn unbalanced again, starting up afresh a market for government bonds would entail significant costs. It has also been suggested that governments should introduce new saving vehicles, such as wage-indexed and longevity-indexed bonds, which would allow for sharing across generations the specific risks pertaining to ageing societies, such as those concerning productivity trends, asset valuation and demographic shocks (Bohn, 2002).

7. Conclusion

The implications of public debt have long been debated. The discussion involved different professions and covered a wide range of topics. This paper offers a bird's eye view of the main issues.

We followed a long thread. We moved from the discussion of the good uses to which public debt can be put in principle to the analysis of its possible misuses. We highlighted the costs of such misuses and stressed the importance of debt sustainability analysis and of effective control mechanisms, both market and rule-based. We concluded with the implications of fiscal rules for debt management.

The public finance tradition justifies government deficits in the face of exceptional circumstances, for financing fixed capital formation and when the economy is under unfavourable macroeconomic conditions. Public bonds are also a relevant instrument for the development of financial markets and the conduct of monetary policy. These factors should not normally justify high debts.

Both theory and experience confirm the intuition of the earlier studies concerning the risk that myopic or opportunistic behaviour by policy-makers results in unsustainable debt dynamics. High public debts can induce distortions and reduce growth. They can determine financial crises with disruptive effects. The analysis of debt sustainability is made problematic by the lack of a fully specified theoretical framework and by practical measurement difficulties. Information problems also hamper the effectiveness of discipline inducing mechanisms. While the market alone seems unlikely to provide adequate incentives, a rule based approach, though used in many countries, is not unproblematic. In general, state-contingent rules are preferable but they are not always workable.

Public debt can be defined in different ways, depending on the sector of reference and the liabilities to be considered. Reference to a gross measure seems most appropriate, however net debt and more comprehensive measures of public liabilities can provide important complementary information. The introduction of fiscal rules may increase the incentives for opportunistic behaviour and even produce adverse effects by distorting the priorities of debt management.

generally inefficient because it is likely to be based on political preferences rather than on market signals and it is likely to create incentives for private firms to lobby for benefiting from public investments. See Gokhale (2002).

Along our journey through the extremely varied land of debt finance, we did not stop at all sites that would have deserved a visit. Where we stopped, we did not stay for long. Even so, we have highlighted many problems which remain open, in spite of a long debate and an extensive literature. While a few years ago some economists worried about the consequences of a disappearing public debt, public debt remains a major policy issue both in developed and in emerging economies. Public debt will remain with us in the future (and so will the unending debate among economists).

REFERENCES

- Aiyagari, S.R. and E.R. McGrattan (1998), "The optimum quantity of debt", *Journal of Monetary Economics*, No. 42, pp. 447-69.
- Alesina, A. (1988), "The End of Large Public Debts", in F. Giavazzi and L. Spaventa (eds.), *High Public Debt: The Italian Experience*, Cambridge, Cambridge University Press.
- Alesina, A. and G. Tabellini (1990), "A Positive Theory of Budget Deficits and Government Debts", *Review of Economic Studies*, No. 57, pp. 403-14.
- Alesina, A. and A. Drazen (1991), "Why Are Stabilizations Delayed?", *American Economic Review*, Vol. 82, pp. 1170-88.
- Alesina, A. and R. Perotti (1995), "The Political Economy of Budget Deficits", IMF, Staff Papers, Vol. 42, pp. 1-31.
- Alesina, A. and T. Bayoumi (1996), "The Costs and Benefits of Fiscal Rules: Evidence from US States", NBER, Working Paper, No. 5614.
- Artis, M. and M. Marcellino (1998), "Fiscal Solvency and Fiscal Forecasting in Europe", CEPR, Discussion Paper, No. 1836.
- Aschauer, D. (1985), "Fiscal Policy and Aggregate Demand", *American Economic Review*, No. 75, March, pp. 117-27.
- Auerbach, A.J. and L.J. Kotlikoff (1987), *Dynamic Fiscal Policy*, Cambridge, Cambridge University Press.
- Auerbach, A.J., J. Gokhale and L.J. Kotlikoff (1991), "Generational Accounts: A Meaningful Alternative to Deficit Accounting", in D. Bradford (ed.), *Tax Policy and the Economy*, Vol. 5, pp. 55-110, Cambridge (Mass.).
- Bach, L. and R.A. Musgrave (1941), "A Stable Purchasing Power Bond", *American Economic Review*, December.
- Balassone, F. and D. Franco (2000a), "Assessing Fiscal Sustainability: A Review of Methods with a View to EMU", in Banca d'Italia (2000).
- ———— (2000b), "Public Investment, the Stability Pact and the Golden Rule", *Fiscal Studies*, Vol. 21, No. 2, pp. 207-29.
- Balassone, F. and M. Francese (2004), "Cyclical Asymmetry in Fiscal Policy, Debt Accumulation and the Treaty of Maastricht", Temi di Discussione, Banca d'Italia, forthcoming.
- Balassone, F., D. Franco and R. Giordano (2004), "Market-induced Fiscal Discipline: Is There a Fall-back Solution for Rule-failure?", *mimeo*.

- Balassone, F., D. Franco and S. Zotteri (2002), "Fiscal Rules: Indicators and Underlying Statistical Frameworks", in Statistical Office of the Republic of Slovenia and Statistical Society of Slovenia (eds.), *Integration of Statistics in the Information Society*, Radenci, Slovenia.
- Balassone, F. and R. Giordano (2001), "Budget Deficits and Coalition Governments", *Public Choice*, Vol. 106, pp. 327-49.
- Balassone, F. and D. Monacelli (1999), "EMU Fiscal Rules: Is there a Gap?", Temi di Discussione, No. 375, Rome, Banca d'Italia.
- Ball, L. and N.G. Mankiw (1995), "What Do Budget Deficit Do?", paper prepared for the Federal Reserve Bank of Kansas City symposium *Budget Deficits and Debt*, Jackson Hole, August 31-September 2.
- Banca d'Italia (ed.) (2001), Fiscal Rules, Rome.
- Barrel, R. and K. Dury (2001), "Will the SGP Ever Be Breached?", in A. Brunila, M. Buti e D. Franco (eds.) (2001).
- Barro, R.J. (1974), "Are Government Bonds Net Worth?", *Journal of Political Economy*, Vol. 82, No. 6, pp. 1095-117.
- Beltrametti, L. (1993), "Una stima della ricchezza pensionistica per l'Italia (1951-1991)", *Rivista Internazionale di Scienze Sociali*, No. 1, pp. 3-15.
- Blanchard, O., J.C. Chouraqui, R.P. Hagemann and N. Sartor (1990), "The Sustainability of Fiscal Policy: New Answers to Old Questions", OECD, Economic Studies, No. 15.
- Blejer, M.I. and A. Cheasty (1991), "The Measurement of Fiscal Deficits: Analytical and Methodological Issues", *Journal of Economic Literature*, Vol. 29, No. 4.
- Bohn, H. (1988), "Why Do We Have Nominal Government Debt?", *Journal of Monetary Economics*, Vol. 21, pp. 127-40.
- ———— (2002), "Government Assets and Liability Management in an Era of Vanishing Debt", *Journal of Money Credit and Banking*.
- Bohn, H. and R. Inman (1996), "Balanced Budget Rules and Deficits: Evidence from the US States", Carnegie Rochester Conference Series on Public Policies, Vol. 45, pp. 13-76.
- Bowen, W.G., R.G. Davis and D.H. Kopf (1960), "The Public Debt: A Burden on Future Generations?", *American Economic Review*, Vol. L, No. 4, pp. 701-06.

- Browning, E.K. (1975), "Why the Social Insurance Budget Is too Large in a Democracy", *Economic Inquiry*, Vol. 13, pp. 373-88.
- Brunila, A., M. Buti and D. Franco (eds.) (2001), *The Stability and Growth Pact: The Architecture of Fiscal Policy in EMU*, New York, Palgrave.
- Buchanan, J.M. (1958), *Public Principles of Public Debt*, Homewood (Ill.), Richard D. Irwin.
- ———— (1976), "Barro on the Ricardian Equivalence Theorem", *Journal of Political Economy*, Vol. 84, pp. 337-42.
- Buchanan, J.M. and J. Roback (1987), "The Incidence and Effects of Public Debt in the Absence of Fiscal Illusion", *Public Finance Quarterly*, Vol. 15, No. 1, pp. 5-25.
- Buchanan, J.M., C. Rowley and R. Tollison (eds.) (1986), *Deficits*, Oxford, Basil Blackwell.
- Buchanan, J.M. and R. Wagner (1967), *Public Debt in a Democratic Society*, American Enterprise Institute.
- ———— (1977), Democracy in Deficit: the Political Legacy of Lord Keynes, New York, Academic Press.
- Buiter, W.H. (1983), "Measurement of the Public Sector Deficit and Its Implication for Policy Evaluation and Design", IMF, Staff Papers, Vol. 30, No. 2, pp. 306-49.
- ———— (1985), "A Guide to Public Sector Debt and Deficits", *Economic Policy*, No. 1.
- Buiter, W.H. and J. Tobin (1980), "Debt Neutrality: A Brief Review of Doctrine and Evidence", in G. von Fustemberg (ed.), *Social Security vs. Private Savings*, Cambridge.
- Buti, M., S. Eijffinger and D. Franco (1998), "Fiscal Discipline and Flexibility in EMU: The Implementation of the Stability and Growth Pact", *Oxford Review of Economic Policy*, Vol. 14, No. 3, pp. 81-97.
- ———— (2003), "Revisiting the Stability and Growth Pact: Grand Design or Internal Adjustment?", CEPR, Working Paper, No. 3692, also European Commission, Economic Paper, No. 180.
- Buti, M., D. Franco and H. Ongena (1997), "Budgetary Policies During Recessions: A Retrospective Application of the Stability and Growth Pact to the Post-War Period", European Commission, Economic Paper, No. 121.

- Buti, M. and A. Sapir (1998), *Economic Policy in EMU: A Study by the European Commission Services*, Oxford, Clarendon Press.
- Calvo, G.A. and P.E. Guidotti (1989), "Indexation and Maturity of Government Bonds: An Exploratory Model", in R. Dornbusch and M. Draghi (eds.), *Public Debt Management: Theory and History*, pp. 52-81, Cambridge, Cambridge University Press.
- Castellino, O. (1985), "C'è un secondo debito pubblico (più grande del primo)?", *Moneta e credito*, No. 149.
- Chalk, N. and R. Hemming (2000), "Assessing Fiscal Sustainability in Theory and Practice", in Banca d'Italia (2000).
- Chase, S. (1943), Where's the Money Coming From?, New York, Twentieth Century Fund.
- Chiancone, A. (1986), Public Debt in the History of Italian Financial Thought, *Finanzarchiv*, No. 2, pp. 253-70.
- ———— (1993), "Political Theories of Public Debt", *Rivista di diritto finanziario e scienza delle finanze*, Year LII, Vol. 3, No. I, pp. 309-23.
- Clarke, P. (1998), "Keynes, Buchanan and the Balanced Budget Doctrine", in J. Maloney (ed.), *Debt and Deficits*, Cheltenham, Edward Elgar.
- Colm, G. and P. Wagner (1963), "Some Observations on the Budget Concept", *Review of Economic Studies*, Vol. 45, pp. 122-26.
- Committee for the Study of Economic and Monetary Union (1989), *Delors Report*, *Report of Economic and Monetary Union in the European Community*.
- Cuckierman, A. and A. Meltzer (1989), "A Political Theory of Government Debt and Deficits in a Neo-Ricardian Framework", *America Economic Review*, No. 79, pp. 713-32.
- De Haan, J., J.E. Sturm and B.J. Sikken (1996), "Government Capital Formation: Explaining the Decline", *Review of World Economics*, No. 132, pp. 55-74.
- De Viti de Marco, A. (1953), Principi di Economia Finanziaria, Torino, Einaudi.
- Diamond, P. (1965), "National Debt in a Neoclassical Growth Model", *American Economic Review*, Vol. LV, December, pp. 1126-50.
- Domar, E.D. (1944), "The Burden of the Debt and the National Income", *American Economic Review*, December.
- Drazen, A. (2000), *Political Economy in Macroeconomics*, Princeton University Press
- ECOFIN (2001), Opinion on the Content and Format of the Stability and Convergence Programmes (Code of Conduct, 2001), available at: http://europa.eu.int/comm/economy finance/about/activities/sgp/codeofconduct_en.pdf

- Economic Policy Committee (2003), *The Impact of Ageing Populations on Public Finances: Overview of Analysis Carried Out at EU Level and Proposals for a Future Work Programme*, available at: http://europa.eu.int/comm/economy finance/epc/documents/2003/pensionmaster_en.pdf
- Einaudi, L. (1948), Principi di Scienza della Finanza, Torino, Boringhieri.
- Elmendorf, D.W. and N.G. Mankiw (1998), "Government Debt", Discussion Paper, No. 1820, Harvard Institute for Economic Research, Harvard University.
- Eltis, W. (1998), "Debt, Deficit and Default", in J. Maloney (ed.), *Debt and Deficits*, Cheltenham, Edward Elgar.
- European Central Bank (2003), Economic Bulletin, November.
- European Commission (1990), "One Market, One Money", European Economy, No. 44.
- ———— (2001), Public Finances in EMU 2001, European Economy, No. 3.
- ———— (2003), Public Finances in EMU, *European Economy*, No. 3, available at:
 - $http://europa.eu.int/comm/economy\ finance/publications/european\ economy/\ 2003/ee303en.pdf$
- Feldstein, M. (1976), "Perceived Wealth in Bond and Social Security: A Comment", *Journal of Political Economy*, Vol. 84, April, pp. 331-76.
- ———— (2004), *Budget Deficits and National Debt*, L.K. Jha Memorial Lecture, Reserve Bank of India, January 12.
- Ferejohn, J. and K. Krehbiel (1987), The Budget Process and the Size of the Budget, *American Journal of Political Science*, No. 31, pp. 296-320.
- Ferguson, J.M. (ed.) (1964), *Public Debt and Future Generations*, Chapel Hill, University of North Carolina Press.
- FitchRatings (2004), Europe's Stability and Growth Pact: Picking up the Pieces. Special Report, 19 January.
- Flemming, J.S. (1988), "Debt and Taxes", *Scottish Journal of Political Economy*, Vol. 35, No. 4, pp. 305-17.
- Franco, D. (1995), "Pension Liabilities Their Use and Misuse in the Assessment of Fiscal Policies", Economic Papers, European Commission, No. 110.
- Franco, D. and M.R. Marino (2004), "The Role of Long-term Fiscal Projections", paper presented at the workshop *Long-term Fiscal Projections*, CentrA, Seville, February.

- Franco, D. and T. Munzi (1996), "Public Pension Expenditure Prospects in the European Union: A Survey of National Projections", *European Economy*, Reports and Studies, No. 3, Brussels.
- Gale, D. (1989), "The Efficient Design of Public Debt", in R. Dornbusch and M. Draghi (eds.), *Public Debt Management: Theory and History*, pp.14-47, Cambridge, Cambridge University Press.
- Gale, W.G. and P.R. Orszag (2002), "The Economic Effects of Long-term Fiscal Discipline", Urban Institute and Brooking Institution Tax Policy Center, Discussion Paper.
- Giordano, R. (2001), "Optimal Debt Maturity under EMU", Temi di Discussione, No. 401, Rome, Banca d'Italia.
- Gokhale, J. (2002), "U.S. Fiscal Policy in an Era of Federal Budget Surpluses", in *The Impact of Fiscal Policy*, Banca d'Italia (ed.), Rome.
- Gramlich, E. (1995), "The Politics and Economics of Budget Deficit Control: Policy Questions and Research Questions", in J. Banks and E. Hanushek (eds.), *Modern Political Economy: Old Topics, New Directions*, Cambridge, Cambridge University Press.
- Griziotti, G. (1917), "La diversa pressione tributaria del prestito e dell'imposta", *Giornale degli Economisti*, March.
- Hagemann, R.P. and C. John (1995), "The Fiscal Stance in Sweden: A Generational Accounting Perspective", IMF, Working Paper, No. 105.
- Hagemann, R.P. and G. Nicoletti (1989), "Ageing Populations: Economic Effects and Implications for Public Finance", OECD, Department of Economics and Statistics, Working Paper, No. 61.
- Hamilton, J.D. and M.A. Flavin (1986), "On the Limitations of Government Borrowing: A Framework for Empirical Testing", *American Economic Review*, Vol. 76, No. 4, pp. 808-19.
- Hansen, A.A. (1941), Fiscal Policy and Business Cycles, New York, Northon & C.
- Haveman, R. (1994), "Should Generational Accounts Replace Public Budgets and Deficits?", *Journal of Economic Perspectives*, Vol. 8, No. 1, pp. 95-111.
- Heller, P.S. (2003), Who Will Pay? Coping with Aging Societies, Climate Change, and Other Long-term Fiscal Challenges, Washington (D.C.), IMF.
- Heller, P.S., R. Hemming and P. Kohnert (1986), "Ageing and Social Expenditures in the Major Industrialized Countries, 1980-2025", IMF, Occasional Paper, No. 47.
- Hills, J. (1984), "What is the Public Sector Worth?", *Fiscal Studies*, Vol. 5, No. 1, pp. 18-31.
- HM Treasury (1998a), The Code of Fiscal Stability, November.

- HM Treasury (1998b), Economic and Fiscal Strategy Report.
- ———— (2002), Core Debt: An Approach to Monitoring the Sustainable Investment Rule, April.
- ———— (2004), The Stability and Growth Pact: A Discussion Paper, March.
- Hoffman, J. (1993), *The Remaking of Europe Employment and the Hidden Debt*, CS First Boston.
- Holcombe, R.G., J.D. Jackson and A. Zardkoohi (1981), "The National Debt Controversy", *Kyklos*, Vol. 34, No. 2, pp. 186-202.
- IMF (1993), World Economic Outlook, October, Washington (D.C.).
- ———— (1996), World Economic Outlook, May, Washington (D.C.).
- ———— (2001), "Fiscal Policy and Macroeconomic Stability", World Economic Outlook, May.
- ———— (2003), "Public Debt in Emerging Markets", World Economic Outlook, September.
- Kilpatrick, A. (2001), Transparent Frameworks, Fiscal Rules and Policy-Making under Uncertainty, in Banca d'Italia (ed.), *Fiscal Rules*, Rome.
- Knight, B. and A. Levinson (1999), "Rainy Day Funds and State Government Savings", *National Tax Journal*, Vol. LII, No. 3, pp. 459-72.
- Kopits, G. and J. Craig (1998), "Transparency in Government Operations", IMF, Occasional Paper, No. 158.
- Kopits, G. and S. Symansky (1998), "Fiscal Policy Rules", IMF, Occasional Paper, No. 162.
- Kotlikoff, L.J. (1984), "Economic Impact of Deficit Financing", IMF, Staff Papers, Vol. 33.
- Kuné, J.B., W.F.M. Petit and A.J.H. Pinxt (1993), "The Hidden Liabilities of Basic Pension Systems in the European Community", CEPS, Working Document, No. 80, November.
- Lamfalussy, A. (1989), "A Macro-Coordination of Fiscal Policies in an Economic and Monetary Union in Europe", in Delors Report (1989), Committee for the Study of Economic and Monetary Union, *Collection of Papers Submitted to the Committee for the Study of Economic and Monetary Union*, pp. 91-125.
- Lane, T. (1993), "Market Discipline", IMF, Staff Papers, No. 40, March, p. 53-88.
- Lerner, A.P. (1943), "Functional Finance and the Federal Debt", *Social Research*, No. 10, pp. 38-51.
- Levin, J. (1993), "The Cash Deficit: Rationale and Limitations", in M.I. Blejer and A. Cheasty (eds.), *How to Measure the Fiscal Deficit: Analytical and Methodological Issues*, Washington (D.C.), IMF.

- Lindbeck, A. (1968), "Theories and Problems in Swedish Economic Policy in the Post-War Period", *American Economic Review*, June Supplement, pp. 1-80.
- Lucas, R.E. and N.L. Stokey (1983), "Optimal Fiscal and Monetary Policy in an Economy without Capital", *Journal of Monetary Economics*, No. 1, pp. 55-99.
- Mackenzie, G.A. (1989), "Are All Summary Indicators of the Stance of Fiscal Policy Misleading?", IMF, Staff Papers, Vol. 36, No. 4.
- Masson, P. and M. Mussa (1995), "Long-term Tendencies in Budget Deficit and Debt", *Budget Deficits and Debt: Issues and Options*, paper prepared for the Federal Reserve Bank of Kansas City symposium *Budget Deficits and Debt*, Jackson Hole, August 31-September 2.
- McGranahan, L. (1999), "State Budgets and the Business Cycle: Implications for the Federal Balanced Budget Amendment Debate", unpublished; Chicago, Federal Reserve Bank.
- Melitz, J. (2002), "Debt, Deficits and the Behaviour of Monetary and Fiscal Authorities", in M. Buti, J. Von Hagen and C. Martinez-Mongay (eds.), *The Behaviour of Fiscal Authorities*, Palgrave.
- Mishan, E.J. (1963), "How to Make a Burden of the Public Debt", *Journal of Political Economy*, Vol. LXXI, No. 6, pp. 529-42.
- Missale, A. (2003), "Optimal Debt Management with a Stability and Growth Pact", mimeo.
- Modigliani, F. (1961), "Long-run Implications of Alternative Fiscal Policies and the Burden of the National Debt", *Economic Journal*, Vol. 71, No. 284, pp. 730-55.
- Musgrave, R.A. and P.B. Musgrave (1984), *Public Finance in Theory and Practice*, New York, Mc Graw-Hill.
- OECD (1982), Government Debt Management Objectives and Techniques, Paris.
- ———— (1993), Government Securities and Debt Management in the 1990s, Paris.
- Pareto, V. (1943), "Sugli effetti dei prestiti e delle imposte e sulla scienza delle finanze (Lettere al prof. Benvenuto Griziotti)", *Rivista di Diritto Finanziario e di Scienza delle Finanze*, pp. 133-36.
- Patinkin, D. (1948), "Price Flexibility and Full Employment", *American Economic Review*, No. 38, pp. 543-64.
- Peach, R. (2001), "The Evolution of the Federal Budget and Fiscal Rules", in Banca d'Italia (ed.), *Fiscal Rules*, Rome.

- Persson, T. and L.E.O. Svensson (1989), "Why a Stubborn Conservative Would Run a Deficit: Policy with Time-Inconsistent Preferences", *Quarterly Journal of Economics*, No. 104, pp. 325-45.
- Persson, T. and G. Tabellini (2000), *Political Economics: Explaining Economic Policy*, The MIT Press.
- Pigou, A.C. (1929), *A Study in Public Finance*, 2nd revised edition, London, Macmillan.
- Pisani-Ferry, J. (2002), "Fiscal Discipline and Policy Coordination in the Eurozone: Assessment and Proposals", paper prepared for the Group of Economic Analysis of the European Commission, April.
- Poterba, J. (1994), "State Responses to Fiscal Crises: 'Natural Experiments' for Studying the Effects of Budgetary Institutions", *Journal of Political Economy*, Vol. 102, pp. 799-821.
- Premchand, A. (1983), Government Budgeting and Expenditure Controls: Theory and Practice, Washington (D.C.), IMF.
- _____(1995), Effective Government Accounting, Washington (D.C.), IMF.
- Puviani, A. (1903), *Teoria dell'illusione fiscale*, reprinted by F. Volpi, Milano, ISEDI, 1973.
- Rostagno, M. (1995), "Il debito previdenziale italiano Il Fondo Pensioni Lavoratori Dipendenti", mimeo.
- Roubini, N. and J.D. Sachs (1989), "Government Spending and Budget Deficits in the Industrial Countries", *Economic Policy*, Vol. 8, pp. 99-132.
- Sabine, B.E.V. (1970), *British Budgets in Peace and War. 1932-45*, London, Allen and Unwin.
- Scarpelli, G. (2002), "Recenti tendenze nella gestione del debito pubblico dei paesi OCSE", *Rivista Bancaria Minerva Bancaria*, No. 2, pp. 63-79.
- Scitovsky, T. (1961), "The Burden of the Debt: Comment", *American Economic Review*, March, pp. 137-38.
- Seater, J.J. (1993), "Ricardian Equivalence", *Journal of Economic Literature*, Vol. XXXI, March, pp. 142-90.
- Seater, J.J. and R.S. Mariano (1985), "New Tests of the Life Cycle and Tax Discounting Hypotheses", *Journal of Monetary Economics*, Vol. 15, March, pp. 195-215.
- Shoup, C.S. (1962), "Debt Financing and Future Generations", *Economic Journal*, December, pp. 887-98.
- Spaventa, L. (1987), "The Growth of Public Debt", IMF, Staff Papers, June, pp. 374-99.

- Sraffa, P. (ed.) (1951), David Ricardo. Works and Correspondence, Cambridge.
- Stella, P. (1993), "Toward Defining and Measuring the Fiscal Impact of Public Enterprises", in M.I. Blejer and A. Cheasty (eds.), *How to Measure the Fiscal Deficit: Analytical and Methodological Issues*, Washington (D.C.), IMF.
- Steve, S. (1972), Lezioni di scienza delle finanze, Padova, Cedam.
- Summers, H.L. (1986), "Debt Problems and Macroeconomic Policies", NBER, Working Paper, October, No. 2061.
- Tabellini, G. (1990), "A Positive Theory of Social Security", NBER, Working Paper, No. 3272.
- Tabellini, G. and A. Alesina (1990), "Voting on the Budget Deficit", *American Economic Review*, Vol. 80, pp. 37-49.
- Tanzi, V. and N. Chalk (2000), "Impact of Large Public Debt on Growth in the EU: A Discussion of Potential Channels", *European Economy Reports and Studies*, No. 2, European Commission.
- Tanzi, V. and D. Fanizza (1995), "Fiscal Deficit and Public Debt in Industrial Countries", IMF, Working Paper, No. 49.
- Ter-Minassian, T. (ed.) (1997), Fiscal Federalism in Theory and Practice, Washington (D.C.), International Monetary Fund.
- Ter-Minassian, T. and J. Craig (1997), "Control of Subnational Government Borrowing", in T. Ter-Minassian (ed.) (1997).
- Tobin, J.M. (1963), Fiscal and Debt Management Policies, Englewood Cliffs, Prentice-Hall.
- Towe, C.M. (1991), "The Budgetary Control and Fiscal Impact of Government Contingent Liabilities", IMF, Staff Papers, Vol. 38, No. 1.
- United Nations (1951), Budgetary Structure and Classification of Government Accounts, New York.
- Van den Noord, P. and R. Herd (1993), "Pension Liabilities in the Seven Major Economies", OECD, Economics Department, Working Paper, No. 142.
- Velasco, A. (1999), "A Model of Fiscal Deficits and Delayed Fiscal Reforms", in J. Poterba, and J. Von Hagen (eds.), *Fiscal Institutions and Fiscal Performance*, Chicago, University of Chicago Press.

- Von Hagen, J. (1992), "Budgeting Procedures and Fiscal Performance in the European Communities", *Economic Papers*, No. 96, Bruxelles, European Commission.
- Von Hagen, J. and I. Harden (1994), "National Budget Processes and Fiscal Performance", *European Economy Reports and Studies*, No. 3, Bruxelles, European Commission.
- ———— (1995), "Budget Processes and the Commitment to Fiscal Discipline", European Economic Review, No. 39, pp. 771-79.
- West, E.G. (1975), "Public Debt Burden and Cost Theory", *Economic Inquiry*, No. 13, pp.179-90.
- Wren-Lewis, S. (2003), "Changing the Rules: Why We Should not Accede to EMU's Current Fiscal Regime", *New Economy*, Vol. 10, No. 2, pp. 73-78.
- Wyplosz, C. (2003), "Fiscal Policy and Macroeconomic Stabilisation in the Euro Area: Possible Reforms of the Stability and Growth Pact and National Decision-Making Processes", in *European Economic Advisory Group Report on the European Economy*.